

Contents lists available at ScienceDirect

Process Safety and Environmental Protection

journal homepage: www.journals.elsevier.com/process-safety-and-environmental-protection

Vanadium – Valuable and toxic element in coal combustion ash: An overview





Lucie Bartoňová^{*}, Helena Raclavská, Jan Najser

ENET Centre, CEET, VSB-Technical University of Ostrava, Tr. 17. listopadu 15, Ostrava-Poruba 708 33, Czech Republic

ARTICLE INFO ABSTRACT Keywords: The paper evaluates all important aspects related to V present in coal and its behaviour during coal combustion Vanadium with particular attention paid to its potential extraction from ash and slag wastes. Due to great environmental Coal combustion benefits of co-combustion of coal and wastes, these alternative fuels and corresponding ash and slag counterparts Ash were also taken into evaluation. Specifically, the paper summarizes V levels in world coals, in other related Physical separation materials (alternative fuels, additives etc.) and in ash and slag residues including V speciation and discussion of Leaching its valence states. Theoretically predicted (equilibrium-based) calculations and species are compared with Wastes experimental results; due to a significant effect on mitigating CO2 emissions, attention is paid also to oxy-fuel combustion conditions. With the aim to extract V out of the ash and slag residues, physical, chemical, and biotechnological separation methods are discussed in detail along with dominant affecting factors. The paper includes also concluding remarks and highlights the most promising trends and limitations thereby providing suggestions for future research.

1. Introduction

Annual production of V is estimated to be around 100.000 t (Bielowicz, 2020; Chinach, 2020); the largest producers are China, South Africa and Russia (Bielowicz, 2020). V is widely used in industrial applications. Due to its high tensile strength, hardness, and fatigue resistance, about 85% of V is used in the production of high-strength ferrous and non-ferrous alloys (Moskalyk and Alfantazi, 2003). According to Moskalyk and Alfantazi (Moskalyk and Alfantazi, 2003) there is still no acceptable substitute for V in the production of aerospace Ti alloys. Other important applications are production of catalysts, inks, dyes, pigments for ceramics and textile industry, batteries, or materials for nuclear reactors (Bielowicz, 2020; Zhang et al., 2019a; Font et al., 2007).

According to European Commission (2020), V belongs to the 2020 list of critical raw materials, i.e., materials that are the most important economically and with a high supply risk – possible interruption in V supply would have harmful consequences on economy (European Commission, 2020). V is the 22-th the most abundant element in earth crust with average concentration of 150 g/t (Fleischer, 1953; Zhang et al., 2019a). As V is not present in high concentrations in minerals, other sources are typically used – it can be recovered as a by-product

during extraction of other components from petroleum residues, metallurgical slags, ore concentrates, waste materials etc. (Moskalyk and Alfantazi, 2003; Vitolo et al., 2000; Bielowicz, 2021).

Combustion residues (ashes or slags) are typical industrial wastes produced in vast quantities throughout the world, e.g. in India (112 Tg/ y), China (100 Tg/y), USA (75 Tg/y), Germany (40 Tg/y), UK (15 Tg/y), Poland (4 Tg/y) etc. (Świetlik et al., 2012; Zhao et al., 2018a). Combustion ashes/slags are therefore intensively studied as a possible source of valuable components, such as char (Badenhorst et al., 2020; Bartoňová, 2015; Nunes et al., 2022; Hower et al., 2017), alumina (Guo et al., 2019), Ge-Ga (Dai et al., 2014a; Zhou et al., 2021), REE (Wagner and Matiane, 2018; Fu et al., 2022; Hower et al., 2020, 2016; Valentim et al., 2019) or Y (Bartoňová et al., 2018; Vassilev and Vassileva, 2020). Unfortunately, papers evaluating combustion ashes as an available source of V are very scarce or nearly non-existent.

During coal combustion, V present in input fuel is redistributed among emissions and solid combustion residues – ashes or slags. Due to medium/low V volatility, environmental research is primarily focused to other (usually more volatile) elements. Therefore, papers dealing with mitigating gaseous emissions are focused predominantly to Hg (Wierońska et al., 2018; Scala et al., 2011; Hower et al., 2005), As (Wierońska-Wiśniewska et al., 2022; Fu et al., 2021), Se (Yu et al., 2022), Pb

https://doi.org/10.1016/j.psep.2023.02.070

Received 5 December 2022; Received in revised form 23 February 2023; Accepted 24 February 2023 Available online 27 February 2023

^{*} Corresponding author. E-mail address: lucie.bartonova@vsb.cz (L. Bartoňová).

^{0957-5820/© 2023} The Authors. Published by Elsevier Ltd on behalf of Institution of Chemical Engineers. This is an open access article under the CC BY-NC-ND license (http://creativecommons.org/licenses/by-nc-nd/4.0/).

(Bartoňová et al., 2019; Zhao et al., 2018b, 2017), Cd (Bartoňová et al., 2020; Fu et al., 2019a), Cu and Zn (Zha et al., 2018, 2020) or Cr (Zhao et al., 2018b; Bartoňová and Raclavská, 2022; Wagner et al., 2021; Fu et al., 2019b).

Hence, due to medium/low V volatility, large proportion of V is bound in combustion ashes/slags, which may hinder their further technological use, handling, storage, or landfilling. According to EPA report from 2011 (US EPA, 2011), V is included in two pollutant groups - "Heavy metals" and "Inorganic pollutants". Exposure to V compounds may cause coughing, irritation of upper respiratory tract, gastrointestinal diseases, or conjunctivitis. The toxicity depends on V speciation and generally increases with increasing valence state - pentavalent V species are the most toxic (Barceloux and Barceloux, 1999; Navarro et al., 2007). Therefore, V is traditionally considered as an element of environmental concern (Wagner and Hlatshwayo, 2005; Vassilev et al., 2014; George et al., 2008; Goodarzi, 2002; Swaine, 1990; Duchesne et al., 2018). Out of this perspective, recovery of V from solid combustion wastes can lead to multiple benefits – providing critical metal while enhancing the perspectives of the combustion wastes for their further technological use.

Due to rapid urbanization and population growth, generation of other than industrial wastes is growing fast as well (Tun et al., 2018; Tun and Juchelková, 2019a). The co-combustion of coal and biomass, municipal solid waste etc. contributes to CO2 neutral conversion and climate change benefits (Xu et al., 2020; Zhou et al., 2014a; Růžičková et al., 2018). It addresses both the depletion of fossil fuels and environmental problems with greenhouse gas emissions; moreover, in case of certain wastes, thermal treatment is taken into consideration if there is a danger of releasing harmful organic pollutants during landfilling or if mechanical recycling is for any reasons problematic (Kucbel et al., 2019; Raclavská et al., 2018). Therefore, it is not surprising that thermochemical conversion of wastes and biomass is one of the most actively studied areas in the world science (Vershinina et al., 2022; Corsaro et al., 2016; Demirbas, 2004). However, different composition and other characteristics of such ash and slag wastes require attention due to their broad variability in dependence of the waste material used

(Růžičková et al., 2018; Vassilev et al., 2015; Demirbas, 2004).

For the aforementioned reasons, this review paper evaluates all important aspects related to V present in coal and its behaviour during coal combustion with particular attention paid to its potential extraction from ash and slag residues. Due to great environmental benefits of cocombustion of coal and wastes, these alternative fuels and corresponding ash and slag counterparts were also taken into evaluation. The scheme of the evaluated research area is depicted in Fig. 1. Specifically, Section 2 summarizes V levels in world coals as well as in other related materials (biomass, additives etc.) and Section 3 deals with thermodynamic equilibrium predictions. Section 4 evaluates concentration of V in ash and slag residues including the effect of operating conditions (e.g., oxy-fuel combustion) and V speciation. Section 5 discussed physical separation methods and their efficiency if applied on ash and slag wastes, while Section 6 provides an overview of chemical and biotechnological ways of V extraction from these materials. Section 7 includes concluding remarks and highlights the most promising trends (as well as the limitations) providing suggestions for the future research.

2. V in coals and other related materials

2.1. V concentrations

Concentrations of V in coal combustion ashes/slags are given not only by operating conditions but namely by V levels in coal; in case of co-combustion with other materials (wastes) V levels therein also affect the resulting concentration in ash/slag.

Examples of V concentrations in coals from world coal regions are summarized in Table 1; concentrations in coal gangue/rejects and limestone (typically used as desulfurization additive) are listed in Table 2.

World averages for low-rank and high-rank coals are 22 and 28 ppm (g/t) V (Ketris and Yudovich, 2009); nevertheless, the data from Table 1 document broad fluctuation (variability) even within 1 seam or coal-field. There are coals containing a few ppm V as well as coals with a few hundred ppm V. Most world coals contain V content below 100 ppm



Fig. 1. Classification diagram describing evaluated research area.

Table 1

V concentrations (in ppm or $\mu g/g$) in coals from world coal regions.

Table	1	(continued)

Region	Coal	Concentrations	Ref.
World	World average	Arithmetic mean	(Bouška and Pešek,
	lignite (3889 coals)	37.28 Control	1999)
		Central range	
		Maximum 843.0	
	World low rank	22 ± 2	(Ketris and Yudovich,
	coal World bard seel	28 ± 1	2009) Kotris and Verderrich
	World total (all	20	(2009)
	coals)		Ketris and Yudovich
	YAY 11 1	0.100 (6	(2009)
	world coals	2–100 (for most coals)	(Swaine, 1990, 2000)
Northern	4 Northern	0.51–11.3 (mean	(Klavins et al., 2009;
Europe	European peats	4.2)	Růžičková et al.,
			2019a)
Australia	35 Gunnedah Basin	6.90–1436.0	(Ward et al., 1999)
	Coals	(mean 153.15)	
	Australian coals	2–279	(Swaine and Goodarzi,
Brazil	7 feed coals	97.54–124.07	(Silva et al., 2010)
	57 Brazilian coals	23–86	(Pires and Querol,
			2004)
	4 coals Santa Catarina coal	71-87 104 (mean)	(Cutruneo et al., 2014) (Oliveira et al., 2012)
Bulgaria	Underground mine	17 ± 7.9	(Eskenazy and
5	coals		Stefanova, 2007)
	Open-pit mine	19 ± 5.9	(Eskenazy and
	Coais Dobrudza deposit	42 ± 20	(Eskenazy, 2007)
	Bobov dol	76	(Vassilev et al., 1994)
	5 raw coals	51-116	(Yossifova, 2014)
Canada	White Wood Mine	4.79–56.83	(Gentzis and Goodarzi,
	Genesee Mine coal	2-83	(Pollock et al., 2000)
	6 feed coals	8.70–18.0	(Goodarzi, 2002)
	Canadian coals	23–300	(Swaine and Goodarzi,
Colombia	7 commercial feed	< 8-32 (mean 18)	(Santos et al., 2022)
	coals	/	
	(exported to		
China	1324 Chinese coals	35.1	(Dai et al., 2012)
	1123 Chinese coals	51.18	a (Bai et al., 2007)
	1266 Chinese coals	35.05	a (Ren et al., 2006)
	1257 Chinese coals	25	a (Tang and Huang, 2004)
	28 coals from 28	46.81 (arithmetic	(Liu et al., 2017)
	Chinese provinces	mean)	
		35.81 (weighted	
	Lvshuidong mine	16.9–195,	(Dai et al., 2014b)
	drillhole	weighted average	
	10 Lening costs	68.4 22,105	(Outerol et al., 2001)
	Jungar basin coals	∠∠–105 10–200, average	(Queroi et al., 2001) (Dai et al., 2008)
	0	27	
	Yudai coal mine	32-201	(Li et al., 2020)
	Jinqi coal mine Haerijao coal	01−398 < dl - 33	(Li et al., 2020) (Li et al., 2019)
India	Meghalaya	25–500 (in coal	(Mukherjee et al.,
	coalfield	ash)	1992)
	Upper Assam coalfield	200–600 (in coal ash)	Mukherjee et al.
	Jammu coalfield	30–150 (in coal	Mukherjee et al.
		ash)	(1992)
Iran	4 Alborz coals	11-16	(Goodarzi et al., 2006)
Africa	nignveia coalfield	23-31	Wagner and Hlatshwavo, 2005
			Wagner and Tlotleng,
			2012)
	Witbank coalfield (no. 4 seam)	39.2	

Region	Coal	Concentrations	Ref.
			(Bergh et al., 2011; Wagner and Tlotleng, 2012)
	Witbank coalfield (no. 2 seam)	27	(Cairneross et al., 1990; Wagner and Tlotleng, 2012)
	Waterberg coalfield	85–150	(Faure et al., 1996; Wagner and Tlotleng, 2012)
Spain	Spanish subbituminous coal	58.2	(Querol et al., 1995)
	4 Puertollano coals	66–265	(Font et al., 2010)
	Puertollano coals (2 mines)	13-89	(Alastuey et al., 2001)
Turkey	143 Turkish coals	5.5–270 (mean 65, median 52)	(Palmer et al., 2004)
	13 Turkish coals	65–219	(Karayigit et al., 2000)
	Soma coals	97 and 163	(Karayigit et al., 2007)
UK	24 UK coals	6.6–93 (average 31.6)	(Spears and Zheng, 1999)
	Parkgate coal	25.1 (median)	(Spears, 2015, 2017)
USA	142 Gulf Coast	4.9–110 (median	(Palmer et al., 2004)
	coals	28)	
	205 Fort Union	1.1–110 (median	
	coals	7.4)	
	Average in U.S. coals	22	(Orem and Finkelman, 2003)
	U.S. coals	0.14–370	(Swaine and Goodarzi, 1995)
	3 Appalachian coals	4.61, 3.63 and 20.54	(Hower et al., 2015)

^a Papers in Chinese reviewed by Dai et al (Dai et al., 2012).

(Swaine, 2000); V concentrations in coal gangue, wastes, rejects or inclusions are generally moderately higher whereas V levels in limestone are lower compared to coals (Table 2). Absolutely the highest V concentration relates to petcoke; therefore, if such material is added to coal during combustion (gasification), the ashes/slags produced contain substantially higher V concentrations (which is undesirable from environmental point of view but advantageous in terms of searching for available V source).

2.2. Modes of occurrence of V

Not only total concentration but also modes of occurrence of V in coal affect its levels in the combustion residues. According to Finkelman et al. (2018), in high-rank coals 65% of V is associated with (alumino) silicates (probably clays) and 35% is organically associated, whereas in low-rank coal the proportion of organic affinity is higher (50%) while silicate association is lower (50%).

In 21 Chinese coals of different rank (from lignite up to anthracite), in addition to aluminosilicate association (R(V-AI) = +0.42), siderophile affinity was also observed (R(V-Fe) = +0.55). As relationship with S was quite low (R(V-S) = +0.21)), it was concluded that V was bound in other Fe-bearing minerals than sulphides/disulphides (Liu et al., 2016). Siderophile character of V was mentioned also in other papers (Vassilev et al., 2005; Wagner and Hlatshwayo, 2005). The decrease in the proportion of organic affinity with increasing rank observed by Liu et al. (2016) agrees with the conclusion of Finkelman et al. (2018).

The results of Finkelman et al. (2018) are based on evaluation of 20 USGS coal samples from Australia, Great Britain, Canada, Brazil, and the USA (Pennsylvania, West Virginia, Kentucky, Alabama, Wyoming etc.) and were obtained by sequential leaching (discussed with the view of other available data, such as optical petrography, SEM with ED XRF, electron microprobe analysis, ion probe analysis, XRD or density fractionation). In case of Kentucky #9 and Illinois #6 coals, X-ray absorption spectroscopy data are available (Maylotte et al., 1981; Huggins and

Table 2

V concentrations (in ppm or µg/g) in coal gangue, rejects, waste materials cocombusted with coal and limestones

Material	Details	Concentrations	Ref.
Coal gangue	Coal gangue from Guqiao Coal Mine,	54.3	(Zhou et al., 2014a)
	China World coal gangue Clarke value-coal gangue Huainan Coalfield China	100–400 135 10.1–83.1 (average 49.5)	(Zhou et al., 2014b) Zhou et al. (2014b) Zhou et al
	Coal gangue Fired gangue (Coal)	77 88 (34)	(2014b) (Querol et al., 2008) Querol et al. (2008) Ouerol et al.
Coal processing	Coal wastes from dumps	203–295 (average 243)	(2008) (Ciesielczuk et al., 2014)
wastes Coal pulveriser rejects	From < 1% S coal From 1% to 2% S coal From > 3% S coal	16 (ppm of ash) 46 (ppm of ash) 55 (ppm of ash)	(Hower et al., 2005) Hower et al. (2005) Hower et al.
Coal inclusions Raw/washed coals	Coal inclusions Bed coals G. Wood, UK Cerrejon, Colombia Harworth, UK	640 ^a 150 ^b Raw/washed 46.5/16.3 24.5/7.75 50.9/33.9	(Yudovich, 2003) (Yudovich, 2003) (Quick and Irons, 2002) Quick and Irons (2002) Quick and Irons (2002)
Petcoke	Petcoke	1560	(2002) (Mukherjee et al.,
	Petcoke from petroleum rafinery Coal (70%)/Petcoke	586–1237 7950 (ash basis,	(Font et al., 2013) (Silva et al., 2013)
	(30%) feed blend Schlams/coal slurry/ petcoke (2:7:1)	11.3% ash) 442	(George et al., 2008)
Agricultural biomass	World reference plant	0.5 ^c	(Vassilev et al., 2014)
	Clarke for angiospermous plants worldwide Clarke for gymnospermous plants worldwide	1.6 0.69	(Vassilev et al., 2014) Vassilev et al. (2014)
	Peanut shell Wheat straw	2.81 7.17	(Zhou et al., 2014a) Zhou et al. (2014a)
	Soybean stalk Corn stalk Sawdust Wood pellets	1.15 2.92 2.64 ca. 2 (from	(Zhou et al., 2015) (Zhou et al., 2016) Zhou et al. (2016) (Guo and Zhong,
	Straw	graph) 2.24	2018) (George et al.,
	Waste wood Biomass mix (green wood, garden waste, straw, roadside grass and manure)	10 6	2010) (Mukherjee et al., 2003; Sloss, 2007) Mukherjee et al. (2003);Sloss (2007)
Coal-Biomass feed blends	Coal – wet saw dust (3:1) Coal slurry – dry saw dust (3:7) Coal-undried bark (1:2)	44.3 90.4 139	(George et al., 2008) George et al. (2008) George et al.

Sludge

Table 2 (continued)

Material	Details	Concentrations	Ref.	
	Paper sludge	5	(Mukherjee et al.,	
	Municipal sewage	24	2003; Sloss, 2007)	
	sludge		Mukherjee et al.	
			(2003);Sloss	
			(2007)	
	Sewage sludge	15.61	(George et al.,	
			2010)	
Limestone	3 High-Cl limestones	1.3, 5.8 and	(Font et al., 2010)	
		15.1		
	Limestone	11	(Córdoba et al.,	
	Limestone slurry	8	2012)	
			Córdoba et al.	
			(2012)	
	2 Czech limestones	< 2.1	(Klika et al., 2001)	
^a Ash from the coal inclusions				

 $^{\rm b}\,$ Ash from the bed coals

^c Worldwide average of all parts of plants

Huffman, 1996, 2004; Huggins et al., 2009, 1997) and agree with the aforementioned sequential extraction ones (Finkelman et al., 2018).

The XAFS results are consistent – revealing two different V associations in coal - mineral and maceral association. The former is attributed to aluminosilicate affinity, probably with V-rich muscovite $K(V^{3+})$, Al)₂AlSi₃O₁₀(OH)₂ (roscoelite) or illite containing V(3 +). In both these cases, V(3+) prevails (Huggins and Huffman, 1996; Huggins et al., 1997). In organic (maceral) association, V(4 +) is more common than V (3 +) that dominates in aluminosilicates (Maylotte et al., 1981; Huggins and Huffman, 2004). It was hypothesized that maceral association of V relates to poorly crystalline oxide V2O4 or oxyhydroxide VO(OH)2 species (Huggins and Huffman, 2004).

3. Thermodynamic equilibrium calculations

Under standard oxidizing conditions and lower temperatures (below 630 K), according to Frandsen et al. (1994), vanadyl sulfates are formed. At ca. > 630 K, VOSO₄(s) decomposes to V₂O₅ (s,l) according to Eq. (1) (Frandsen et al., 1994):

$2 \operatorname{VOSO}_4(s) \leftrightarrow \operatorname{V_2O_5}(s,l) + 2 \operatorname{SO}_2(g) + \frac{1}{2} \operatorname{O}_2(g)$	(1	1])
----------------------------------------------------------------------------------------------------------------------------------------	----	---	---	---

Between 1350 and 1645 K, V2O5 (s,l) is converted to VO2(g) (which is stable at 1640 K) according to Eq. (2) (Frandsen et al., 1994):

$$V_2O_5(s,l) \leftrightarrow 2 VO_2(g) + \frac{1}{2}O_2(g)$$
 (2)

Under reducing conditions, V also forms O-species but V₂O₅ (s,l) is converted to VO₂ at 1500–1750 K, i.e. at higher temperatures (Frandsen et al., 1994).

As the only gaseous form of V in these equations is VO₂, it could be deduced that at oxidizing conditions V is more volatile than at the reducing ones, which is in agreement with the conclusion of Yan et al. (2001a) that unlike many other trace elements (Mn, Sb, Zn etc.), only V, As and Sn exhibited increased volatility under oxidizing conditions (vs. reducing ones), whatever the coal type (i.e., in both high-ash and low-ash coals). In this study, the difference in temperature for the same V volatility in oxidizing and reducing conditions was ca. 100 °C as well. It also agrees with the conclusions of Duchesne et al. (2018) that in combustion systems V is more volatile than in gasification ones. As higher oxidation states of V (that are more toxic) are expected during combustion (vs. gasification), it might result also in environmental consequences.

According to Yan et al. (2001b), V belongs to elements with high affinity to O, which has the order:

As > Se (Co, Cr, Mn, Pb, Sb, Te, V)> Ni > Sn > Zn >> (Cu, Hg) > other trace elements (Cd, Tl).

V forms O-combined species even at low mole fraction of O $(0\approx 0.16)$; with increasing amount of O, more V-O species appeared. The

(2008)

most abundant are V-Ca (or Fe) oxides, such as FeV_2O_6 (s), CaV_2O_6 (s), CaV_2O_7 (s), $Ca_3V_2O_8$ (s), FeV_2O_4 (s), etc. (Yan et al., 2001b). It is consistent with V species predicted by George et al. (2008, 2010) – VO₂, V_2O_5 , FeO.V₂O₅, CaO.V₂O₅, and 2CaO.V₂O₅. Predicted formation of V_4O_{10} was not observed experimentally, which was explained by kinetic constraints.

In addition to the effect of oxidizing or reducing atmosphere, there is also an effect of the rest of the atmosphere, which means that not only % O₂ but also traditional (conventional) or *oxy-fuel atmosphere* could have as effect on V volatility (Jano-Ito et al., 2014). The comparison of temperatures where more than 50% V was volatilized revealed that this temperature was lower at air-fired combustion (> 660 °C) than at oxy-fuel combustion with 55.9% and 72.1% gas recycling (> 690 °C and > 670 °C). However, this difference was not attributed to replacement of N₂ by CO₂; rather, it was explained by larger concentration of S (due to gas recycling) that bind Ca leaving its lower amount available for the formation of Ca-V oxides which results in more abundant V₂O₅ formation (Jano-Ito et al., 2014). V interacts also with Cl, the temperature of VOCl₃ occurrence is lower at oxy-fuel combustion (100–200 °C) than at air-combustion (ca. 500 °C) (Jano-Ito et al., 2014).

4. V in ash and slag wastes

4.1. Concentrations of V in bottom ash (BA)/slags, fly ash (FA) and desulphurization residues

World averages for V concentrations in brown-coal and hard-coal ashes are 140 and 170 ppm (Ketris and Yudovich, 2009). V concentration ranges in FAs from Australia, Europe, and Canada-USA are 49–274 (median 143) ppm, 154–514 (median 202) ppm and 49–3120 (median 413) ppm, respectively (Riley, 2007; Sloss, 2007).

Elevated V content was observed in industrial FA originated from cocombustion of coal, tire-derived fuel and petcoke (1400 ppm) (Hower et al., 2001); extremely high concentration of V in FBC FA was obtained if petroleum coke was used as input fuel - 5473 ppm V (González et al., 2009). Co-combustion of coal and solid recovered fuel (SRF) containing plastics, wood, paper, fabric, rubber etc. produced FAs with V levels similar to those of conventional coal combustion - FAs from hard coal and SRF contained 103 and 175 ppm V (Wasielewski et al., 2020) or 353 ppm (Wojtaszek et al., 2021), FA originated from lignite and SRF contained 233 ppm V (Bartoňová et al., 2010). World reference for biomass ash (calculated as worldwide average for all parts of all plants and 4.9% ash yield) is 10 ppm (Vassilev et al., 2010, 2014); Clarke value for plant ash is 61 ppm V (Vassilev et al., 2014). V concentrations in desulphurization products have been published in literature as well. European (mostly from Germany), Japanese, and USA FGD gypsums contained 8-16, 21 and 10-12 ppm V, respectively (Berland et al., 2003; Sloss, 2007). V concentrations in FGD sulfate and sulfite were 8 and 15 ppm (Hower et al., 2005); in gypsum and gypsum slurry it was 7 and 5 ppm V (Córdoba et al., 2012). All these values are substantially lower than V contents in ashes or slags (mentioned above).

There are abundant papers reporting V concentrations in BA, slag, and FA samples from coal (co-) combustion; typically, V levels in FAs are moderately higher than those in BAs/slags (and than those in coals). Fig. 2 depicts *enrichment factors (EF)* of V in FA compared to BA/slag (within the same combustion unit) and it was created using the literature data (top to bottom) (Font et al., 2010; Silva et al., 2010; Dahl et al., 2009; Silva et al., 2014; Parzentny and Róg, 2020; Córdoba et al., 2012; Querol et al., 1995; Tang et al., 2013; Hower et al., 1996; Wu et al., 2019; Chen et al., 2019; Karayigit et al., 2007; Spears and Martinez-Tarrazona, 2004; Silva et al., 2010; Chen et al., 2019; Narodoslawsky and Obernberger, 1996).

In most combustion tests, EFs are > 1 indicating V enrichment in FA vs. BA/slag. The only exceptions are two white columns (Narodoslawsky and Obernberger, 1996; Chen et al., 2019) related to grate-fired units; the lowest value in the whole diagram (Narodoslawsky and



Fig. 2. Enrichment factors of V in FA vs. BA/slag. *Numbers in parentheses indicate the number of FA-BA/slag pairs evaluated (and averaged).

Obernberger, 1996) relates to the test operating without coal – therein, only bark and wood were combusted in grate-fired facility. In contrast, it is interesting that the 4 highest EFs relate to units operating with coal and other materials, such as petcoke (EF=1.82) (Font et al., 2010), oil (EF = 1.52) (Silva et al., 2010), forest residue (EF = 1.47) (Dahl et al., 2009), or switchgrass (EF = 1.38) (Silva et al., 2014). Enrichment factors for normal PCC or FBC of (sole) coal are within the range of 1.09–1.37; despite combusting different types/ranks of coal in various units, the EF values are quite similar.

Due to CO_2 neutral conversion and climate change benefits, the replacement of certain part of coal (gangue) by biomass during the combustion is intensively studied (Zhou et al., 2014a, 2015, 2016; Guo and Zhong, 2018). General conclusion of all these studies is decreased V percentage in FA (vs. BA) during combustion due to adding biomass. Even if V percentages of gaseous emissions remained low (ca. 1-2%) during combustion of coal gangue and soybean stalk, peanut shell, or sawdust (or their blends with the gangue), V proportion in BA during combustion of sole biomass was always higher than that in sole gangue in case of soybean stalk and sawdust by ca. 5-10%, in case of peanut shell even by ca. 25%. Although the effect of the overall distribution of the ash particles during the combustion experiment cannot be neglected, higher levels of alkali and alkali earth elements were supposed to play a dominant role (Zhou et al., 2014a, 2015). Co-combustion of anthracite with wood pellets in fluidised bed have led to similar conclusion adding wood pellets inhibit volatilization of V (Guo and Zhong, 2018).

4.2. Distribution of V within ESP/baghouse rows

V concentrations in 3 (or more) rows of the same ESP or baghouse are for various combustion tests shown in Fig. 3. The diagram was created using the literature data (left to fight) (Mardon et al., 2008; Tang et al., 2013; Santos et al., 2022; Santos et al., 2022; Santos et al., 2022; Santos et al., 2022; Kostova et al., 2016; Kostova et al., 2016).

V levels in these FAs are mostly a few hundred ppm, only in some individual cases they exceed 400 ppm. In most combustion tests, V levels gradually increase with increasing number of the row (i.e., in the last rows its content is typically higher). FA from the last rows contain more V; hence, such last-row ashes should be more suitable when searching for the cheap V source; nevertheless, amount of FA collected at the 3rd or later rows is rather low in comparison with FA from the 1st row which is more abundant.

This general trend (V increase in later rows FAs) is valid also for cocombustion of 70% pulverized coal + 30% pet coke (in Kentucky power station) (Silva et al., 2013). FA from the 1st row was not evaluated (due to difficulties with the FA collection), in the 2nd and the 3rd sections the V levels were 30,800 ppm and 35,600 ppm V -these concentrations are high due to high V content in 70% coal/30% petcoke feed fuel -7,



Fig. 3. Concentrations of V in individual rows of ESP or baghouse filters. *Values plotted herein for the 3rd row are in fact mixtures of the 3rd and the 4th row FAs.

950 ppm (ash basis; 11.3% ash). Increasing concentration of V (and other pollutants) (Czech et al., 2020) with decreasing particle size of FA is undesirable from environmental point of view because the finest PM particles can be distributed to large surrounding areas releasing toxic pollutants there (Kantor et al., 2021; Švédová et al., 2019).

4.3. Influence of combustion procedure and operating parameters

Relative enrichment (RE) of V in FA (vs. coal) compared for 3 *types of industrial combustion units* with grate fired, pulverized coal and fluidised bed boilers revealed moderate increase in RE in FA in the order: grate-fired boiler (700 °C) < pulverized coal boiler (1400 °C) < fluidised bed boiler (950 °C) – all these values ranged from ca. 0.7 to ca. 0.8 documenting in all cases V depletion in FA vs. coal (Chen et al., 2019). Apart from the effect of the different V associations in the studied coals, interestingly, this sequence did not follow the combustion temperature increase that was in the different order: grate-fired boiler (700 °C) < fluidised bed boiler (950 °C) < pulverized coal boiler (1400 °C). Since V is not a typical volatile element and generally tends to remain in condensed fractions, it might rather reflect the overall redistribution of mineral matter. Another possible effect relates to unburned carbon levels in these three industrial ashes because the RE value was calculated according to the traditional formula:

$$RE = \frac{(c_i)_{ash}}{(c_i)_{coal}} \bullet \frac{(\% ash)_{coal}}{100}$$

where c_i are the concentrations of the i-th elements (herein of V) in ash and coal. The $(\%ash)_{coal}$ relates to the ash content in the coal and 100 in the denominator is used for the approximate ash content in the evaluated ash or slag. Even if unburned carbon levels in these samples were not published (Chen et al., 2019) it can be presumed that unburned carbon levels (generally) decrease in the order: grate fired boiler > pulverized coal boiler > fluidised bed boiler. Hence, as the RE differences among the boiler types are quite low (0.7–0.8) it could be hypothesized that high unburned carbon levels in the ash could "dilute" V contents there leading to moderately lower RE values (in comparison with RE in the ash containing negligible amount of unburned carbon).

The comparison of *industrial combustion and household stove combustion* (Pędziwiatr et al., 2021) led to the conclusion that ash and soot samples from the household stoves contained lower V concentrations (< 92 ppm) in comparison with industrial BA and FA samples (140–248 ppm), probably due to wood that was co-combusted with bituminous coal in household stove (co-combustion of wood or other biomass types with coal is quite common practice) (Růžičková et al., 2019b).

4.3.1. Oxy-fuel combustion and effect of water

Oxy-fuel combustion. As coal combustion is an important contribution to anthropogenic greenhouse gas (CO₂), modern technologies mitigating CO₂ emissions are intensively studied (Mikulčić et al., 2022). One of the most feasible options in the mid-term horizon is oxy-fuel combustion (Senneca et al., 2017). In comparison with traditional air-combustion, coal is burned usually in a mixture of oxygen and recycled flue gas, which enables to attain high CO₂ concentration in exhaust gas (Davidson and Santos, 2010).

Experimental results of co-combustion of bituminous coal and sawdust under air (21%O₂/79%N₂) and oxy-fuel combustion conditions (21%O₂/79%CO₂) revealed practically no effect of changed atmosphere. Zhou et al. (2020) observed nearly the same ratios of V in FAs at both atmospheres for 800 °C, 900 °C, and 1000 °C combustion experiments (ca. 40-48% according to the temperature); the volatilization ratios remained also unchanged - in both atmospheres it was 2-3%. In the study of Oboirien et al. (2014), the V concentrations in 900 °C BAs generated at 21%O2/79%N2 and 21%O2/79%CO2 atmospheres were exactly the same. The results of Jano-Ito et al. (2014) (described in detail in Section 3) and Zhou et al. (2020) or Oboirien et al. (2014) are quite similar; nevertheless, moderately lower V volatility under oxy-fuel combustion conditions (vs. air combustion) concluded by Jano-Ito et al. (2014) was not observed by Zhou et al. (2020) and Oboirien et al. (2014). This minor difference might be brought about by mixing pure gases by Zhou et al. (2020) and Oboirien et al. (2014), while Jano-Ito et al. (2014) recycled the whole flue gas including S, Cl, H₂O etc.; the elevated concentrations of S, Cl, and H₂O might result in minor decrease in V volatility.

In addition to oxy-fuel combustion, there are also other promising technologies with environmental benefits, such as IGCC (combined optimally with CO₂ capture) (Ordorica-Garcia et al., 2006), co-firing with low C or C-neutral fuels (biomass (McIlveen-Wright et al., 2007), H₂ or ammonia (Xu et al., 2022) or waste-derived fuels (Mikulčić et al., 2016)) or carbon capture technologies (e.g., chemical looping combustion using particulate solid oxygen carrier (Coppola and Scala, 2021) or using mineral carbonation technology (Coppola et al., 2022)). As V behaviour and the ash characteristics are expected to be different and highly variable, more research attention is needed in this field.

Effect of water. Effect of water (moisture) on behaviour of trace elements is generally more pronounced in case of oxy-fuel combustion due to flue gas recycling (thereby increasing levels of H_2O) (Roy et al., 2013) or if coal-water slurry is combusted (Vershinina et al., 2016, 2020; Glushkov et al., 2016). When waste materials containing higher moisture levels are co-combusted with coal, such as sewage sludge, agricultural or forest residues – then different drying methods might also have certain effect (Raclavska et al., 2011; Tun and Juchelková, 2019b).

Papers on effect of water (e.g., during coal water slurry combustion)

on behaviour of elements are extremely scarce or nearly non-existent. In case of Cr, Mn, and Ni, the bottom emission stream was higher in coalwater slurry fuel than in case of pulverized coal combustion, which was attributed to lower temperatures during coal-water slurry fuel combustion and possibly also due to larger agglomerates formed from coalwater droplets (Nodelman et al., 2000). Nevertheless, V was not included in this study. Increase of H₂O content in MSW from 0 to 65 wt% during the pyrolysis at 650 °C led to increase of V content in solid phase (by ca. 4 mg/kg) while its levels in oil and liquids remained low (< 1 mg/kg) and nearly the same regardless the water level (Raclavská et al., 2015). Results from two multi-burner coal-water slurry gasification plants (with ca. 60 wt% coal in slurry containing 9.1 and 9.6 ppm V) revealed lower V levels in coarse slags (114 and 116 ppm V) and higher V levels in fine slags (152 and 155 ppm V) (Jiang et al., 2021).

Higher water levels influence also the catalytic effect of V_2O_5 on oxidation of SO_2 to SO_3 during coal combustion – higher H_2O levels increase the formation of SO_3 ; the most effective catalyst was V_2O_5 while Fe₂O₃ and CuO exhibited somewhat lower effect (Duan et al., 2015).

Moreover, according to Galetz (Galetz, 2015), combined effect of low-quality fuels containing high V concentrations (oil sands, refinery residues etc.) and reducing atmosphere with high levels of CO, water and H_2S can accelerate the corrosion of boilers or heat exchangers if made from low Cr steel.

4.3.2. Combustion additives

Silica-based additive used for the inhibition of ash slagging and fouling (Low et al., 2015) in industrial pulverized coal-fired boiler (1300 °C) provided noticeable reduction in enrichment across all boiler positions not only for V but also for As, Cr, and Mn. Addition of the silica-based additive slightly increased the tendency of V to remain in the gas phase. Thermodynamic equilibrium predictions on V states during coal combustion at 1300 °C revealed significant increase of gaseous (VO₂) fraction if silica-based additive was added. In both cases (with or without silica-based additive), the most abundant solid phase was (CaO)₃(V₂O₅). Formation of Ca-V species has been mentioned also in other studies (Folgueras et al., 2007; Díaz-Somoano et al., 2006) which could be important in terms of using traditional Ca-bearing additives (limestone, dolomite, lime etc.). Calcium vanadium oxide hydrate (Ca₂V₂O₇. 2 H₂O) was identified also in ash from fluidised bed combustion of 100% petcoke where limestone was used as desulphurization additive (Jia et al., 2002). However, in FBC (850 °C) the correlation coefficients V-Ca in BAs from 4 combustion tests with CaCO₃ or CaO added to bituminous coal or lignite were in all cases negative (R was from -0.3 to -0.5) (Bartoňová and Klika, 2014), which is in agreement with the literature results described in the next section (Bielowicz, 2020; Spears and Martinez-Tarrazona, 2004; Singh et al., 2011; Huggins and Huffman, 1996; Huggins et al., 1997). Therefore, it could be hypothesized that synergetic effect of more factors leads to the formation of Ca-V compounds. The effect of Ca-bearing minerals (in this context) is probably enhanced by organic or other volatile V association in the fuel (coal, petcoke etc.). And, the effect of the combustion temperature is important as well because if V is firmly bound in aluminosilicates, low combustion temperature (e.g., in fluidised bed combustion) is not sufficient for V volatilization thereby hindering its interaction with Ca-bearing additives.

4.4. V speciation in ash and slag wastes

As mentioned in Section 2.2., V in *coal* is associated predominantly with aluminosilicates and organic matter (macerals) (Finkelman et al., 2018; Maylotte et al., 1981; Huggins et al., 2009).

Relationships of V with major elements in *coal ashes* were evaluated (i.a.) by Singh et al. (2011), Spears and Martinez-Tarrazona (2004) or Bielowicz (2020). Correlation coefficients calculated for V-Al₂O₃ relationship in coal ash by Singh et al. (2011), Spears and

Martinez-Tarrazona (2004) and Bielowicz (2020) were + 0.89, + 0.54 and + 0.40, respectively. Despite quite big differences in these values, all these three R values are statistically significant. As critical value of correlation coefficient (for 2-tailed probability) decreases with the increase of the number of evaluated data, the critical value for the 9 measurements of Singh et al. (2011) is 0.798 ($\alpha = 0.01$) or 0.666 $(\alpha = 0.05)$, for the data of Spears and Martinez-Tarrazona (2004) it is 0.39 ($\alpha = 0.01$) and for the 28 samples of Bielowicz (2020) it is 0.374 ($\alpha = 0.05$). These results document at least partial association of V with aluminosilicates in the studied coal ash samples. Moreover, high R(V-K) calculated by Spears and Martinez-Tarrazona (2004) is also consistent with previously published V association with illite or roscoelite in coal (Huggins and Huffman, 1996; Huggins et al., 1997). In some FAs, siderophile character of V has also been observed. In Eggborough FA, Spears and Martinez-Tarrazona (2004) based this conclusion on the correlation coefficient calculation where R(V-Fe) = +0.61 (critical value for $\alpha = 0.01$ is 0.39), while Vassilev and Menendez (2005) based it on V enrichment in heavy-mineral fraction (heavy concentrate with $\rho > 2.89 \text{ g/cm}^3$).

Unlike ashes originated from coals, BAs prepared at 500 °C (2 h) from 8 *biomass* samples (beech wood chips, corn cobs, marine macroalgae, weathered plum pits, rice husks, switchgrass, sunflower shells, and walnut shells) did not reveal significant association of V with silicates or Fe (Vassilev et al., 2014). In contrast to coal ashes, V exhibited the affinity to chlorides, sulphates, Na, Cr, Sr, U, and Ni. The strongest association was observed with S (R^2 (V-S) = 0.99) (Vassilev et al., 2014).

In co-combustion of pulverized coal (70%) and petcoke (30%), the XPS binding energy of V was primarily attributed to V(5 +) (Vassilev et al., 2014). HR-TEM/EDS/SAED documented (in FA) multiwalled nanotubes encapsulating V. In ultrafine particles, V-spinels were incorporated into the magnetite structure (replacing Fe^{2+} and Fe^{3+} ions in crystal lattice) (Silva et al., 2013). It could correspond with quite high R (V-Fe) = +0.61 (statistically significant value for $\alpha = 0.01$ is 0.39) in Eggborough PCC FA (Spears and Martinez-Tarrazona, 2004). Results presented by Izquierdo et al. (2007) for 2 FAs from Spanish power plants document practically no effect of % petcoke added to coal during co-combustion on V speciation - in both FAs (4% and 24% petcoke), the prevalent form of V was vanadate VO_4^{3-} with minor proportion (<30%) in lower oxidation states (mainly V(4 +)). In contrast, the effect of combustion/gasification atmosphere on V speciation is more significant (Duchesne et al., 2018). Synchrotron-based XAS revealed that petcoke gasification products contain 79–95% V(3 +) and the rest is V(4 +)while combustion products contain 40–71% V(5 +) and 12–60% V(4 +) (Duchesne et al., 2018), NaVO₃ (Bacci et al., 1983) and vanadyl sulphate (Huffman et al., 2000) were identified at oil-fired power plants.

Details on V speciation are important not only in terms of toxicity or possible V extraction. There is also an effect on corrosion and fouling of combustors that is attributed mainly to V(5 +) while V(4 +) and V(3 +) are considered innocuous (Duchesne et al., 2018).

5. Physical separation

Physical separation methods the most widely applied on coal combustion wastes are particle-size fractionation, magnetic separation, density separation, flotation, electrostatic separation etc.

5.1. Particle-size fractionation

In case of 2 industrial coal combustion FAs (390 and 270 ppm V), the highest V concentrations within particle size fractions were observed in $< 25 \ \mu m$ fractions (480 and 320 ppm V) (Hower et al., 1999). The finest and the coarsest fractions of FA (with bulk 507 ppm V concentration) contained 1236 and 227 ppm V (Lanzerstorfer, 2018); particle-size fraction of another coal FA (with 496 ppm V) revealed the enrichment of V in the finest fraction (596 ppm V) and the depletion in the coarsest one (120 ppm) as well (Mardon and Hower, 2008). Size fractionation

applied on the non-magnetic residue of coal FA (after magnetic separation) is also consistent with the aforementioned trend – the highest V concentration (435.9 ppm) was obtained in the finest fraction, the coarsest fraction contained substantially lower V content (136 ppm) (Querol et al., 1995). The same particle-size distribution trend in FA was observed even if peat was co-combusted with forest residues in fluidised bed boiler (the highest V content 150 ppm V relates to < 0.074 mm fraction). In this fraction, around 94% of V was present. In BA, the highest level of V was in 0.25–0.5 mm fraction (also ca. 150 ppm) and it decreased with increasing particle size to below ca. 20 ppm V (Dahl et al., 2009).

Overall, literature data on particle-size distribution of V in FA provide a clear and consistent trend – enrichment of V in the finest FA fractions.

5.2. Magnetic separation

Amount of magnetic concentrate separated from FA mostly fall within the range of 0.5%– 18.1% (Vassilev et al., 2004). The most abundant magnetic minerals present in magnetic fractions are magnetite, hematite, ferrian spinel, Ca- and Ca-Mg spinels with some minor occurrence of maghemite, martite, mushketovite, wustite, ilmenite, chromite, native Fe, Mn ferrite, ferrosilicon, Fe hydroxides and Fe silicates (Vassilev et al., 2004) (and the references therein).

Comparison of V levels in magnetic and non-magnetic fractions are summarized in Table 3. Results summarized in Table 3 are quite inconclusive: there are samples exhibiting V enrichment in magnetic fractions (Santos et al., 2022; Cornelius et al., 2021), in non-magnetic fractions (Querol et al., 1995) or showing no clear trend (Vassilev et al., 2005; Strzałkowska, 2021; Myazin et al., 2018). These differences could be brought about by different association of V in coal combustion wastes and/or by different separation techniques used in evaluated papers.

According to Myazin et al. (2018), approximate minimum commercial content of a valuable component for V is 500 g/t (ppm or μ g/g). Comparing this value with data in Table 3, the only magnetic concentrate meeting this requirement is magnetic fraction prepared from heavy concentrate of the only 1 FA sample (656 ppm V on air-dried basis) (Vassilev et al., 2005). It is worth mentioning in this context that another FA from the same thermal power station treated in the same way provided only 273 ppm V – which clearly documents the effect of input coal (ash) characteristics (Vassilev et al., 2005). Quite high V levels were reported also by Santos et al. (2022) for 2 magnetic concentrates of 2 ESP FAs – 420 and 372 ppm V. Enrichment factor calculated for V concentration in magnetic concentrate (vs. bulk FA) for 5 Spanish FAs was 1.0 showing no enrichment (Vassilev et al., 2004).

Therefore, the combination of more methods is generally needed if greater enrichment is required.

5.3. Density separation

Density fractionation from slag (155.2 ppm V) and FA (208.3 ppm V) from the same power station provided the highest V content (160 ppm) in 2.4 - 2.8 g/cm³ fraction of slag and 2.4 - 2.8 g/cm³ fraction of FA (229.9 ppm V), which could be given by prevalent aluminosilicate affinity in these samples (Querol et al., 1995).

Average concentration of V in 5 heavy concentrates prepared by sink-float technique in bromoform ($\rho = 2.89 \text{ g/cm}^3$) from 5 FAs generated from 4 Spanish thermal power stations was 256 ppm. As average concentrations of V (ash basis) in these original FAs were 201 ppm, V was enriched in heavy concentrates, which corresponds with predominant siderophile character (Vassilev and Menendez, 2005).

5.4. Char concentrates (from ash) and electrostatic separation

There are numerous procedures used for the separation of carbon concentrates from coal ash, such as triboelectrostatic separation, density separation, oil agglomeration, fluidization or vibrational separation etc. (Lv et al., 2022). Char concentrates separated manually (when dry) from two > 0.1 mm FAs provided 111 and 261 ppm V; the comparison with V levels in bulk FAs (266 and 316 ppm) indicates depletion of V in char from Turkish FAs (Vassilev et al., 2005). In 5 Spanish FAs, average V enrichment factor in carbon concentrates (vs. V content in bulk FAs) was 1.1 showing moderate enrichment.

Table 3

V concentrations (in ppm or $\mu g/g$) in magnetic and non-magnetic fractions of coal combustion ashes/slags.

Combustion	Separation	Ash/slag	Magnetic fraction	Non-magnetic fraction	Ref.
1050 MW European power station	Magnetic separator with ultrasonic bath (in acetone)	FA (from bituminous coal) Slag (from bituminous coal)	227 150	237 160	(Querol et al., 1995)
PCC thermal power station	Dry separation	FA1 FA2	273 ^a 656 ^a	423 ^b /111 ^c 198 ^b /261 ^c	(Vassilev et al., 2005)
Thermal power plant in Eastern Transbaikalia	Laboratory magnetic separation at i)10 kA/m ii)30 kA/m	i)FA1 FA2 ii)FA1 FA2	70 5.0 60 30	60 20 60 30	(Myazin et al., 2018)
Power plants in Poland, bituminous coals	Manual magnetic separator	Slag 1 Slag 2	182.9 174.2	272.6 157.7	(Parzentny and Róg, 2020)
Arnot power station (South Africa) Energetic coal combustion (power	Wet separation in water Dry manual separation using magnet	FA Class F FA Class C FA	126 204 274	119 235 208	(Cornelius et al., 2021) (Strzałkowska, 2021)
PCC combustion at 1500 °C, Colombian coal	Manual wet separation (ferrite magnet)	BA 1 Economizer ash 1 ESP FA 1 BA 2	274 228 236 420 205	208 204 159 229 192	(Santos et al., 2022)
444 MW Kentucky power plant (bitum. coal +petcoke)	2-step separation (dry Eriez separator + wet in water)	Economizer ash 2 ESP FA 2 FA ESP 1st row FA ESP 2nd row	229 372 6159 ^d 5635 ^d	166 239 8188 ^d 8079 ^d	(Henke, 2005)

 $^{\rm a}\,$ Magnetic fraction from heavy (>1 g/cm^3) fraction

^b Light fraction ($< 1 \text{ g/cm}^3$)

 $^{\rm c}\,$ Char concentrate (manually separated from >0.1 mm fraction)

 $^d\,$ After ashing at $750^\circ C$

Electrostatic separation applied on 2 coal combustion wastes (at U = 10 kV) provided for the first sample moderate enrichment in the conductors' fraction and enrichment in the dielectrics fraction for the second one. When U = 20 kV was used, in case of both these samples, the highest V content was observed in the intermediate (semiconductors) fraction. Nevertheless, V content in none of these fractions did not exceed 200 ppm (Myazin et al., 2018).

Thus, overall, individual physical–separation techniques typically do not lead to significant V enrichment that could be sufficient for commercial utilization of coal combustion wastes as a cheap source of V. Therefore, the combination of these techniques is recommended optimally after evaluation of input fuel characteristics. Nevertheless, the "pre-concentrate" thereby obtained might become an advantageous input material for further application of chemical / biotechnological methods (that are discussed in detail in Section 6). And at the same time, such techniques could be used in order to decrease V levels in ashes/ slags. Utilization perspectives of fractions depleted in V (and other trace elements) in terms of their storage, handling, landfilling or further technological use are thereby better.

6. Chemical extraction from ash and slag wastes

Methods of elemental extraction from ash and slag wastes include the following approaches (Yermagambet et al., 2018; Vitolo et al., 2000; Ye et al., 2012):

- i) Acid leaching (of basic or amphoteric compounds using solutions of H_2SO_4 , HCl etc.)
- ii) Alkali leaching (of acidic or amphoteric compounds using solutions of NaOH, NH₄OH etc.)
- iii) Extraction under severe conditions (high temperature and pressure, using aggressive chemicals, roasting etc.)
- iv) Biotechnology leaching using microorganisms.

6.1. Leachability in water

In general, leached amount of elements in water is given by their concentration in ash, mode of occurrence in ash, operating conditions during combustion, adsorption/desorption processes, redox conditions and namely by pH (Vassilev et al., 2020).

V leachability in water from *coal combustion* ashes/slags is generally low. Proportions of V leached from 9 Bulgarian coal FAs (by water) was within the range of 0.1–1.0% (Vassilev et al., 2020). Around 0.1–4% V water leachable fraction from coal FAs reviewed by Izquierdo and Querol (2012) is also quite low and consistent with other authors (Lieberman et al., 2020; Pires and Querol, 2004; Querol et al., 2000; Spears, 2013). However, co-combustion of coal with wastes or even the combustion of biomass without coal might substantially elevate water leachability of V. High V solubilities in water was obtained from biomass ashes (prepared at 500 °C for 2 h) where V was preferably associated with water-soluble fraction – sulphates and chlorides (Vassilev et al., 2014). Therefore, it is recommended to remove the water-soluble phases prior to such ash is used (Vassilev et al., 2014).

When low and high amount of *petcoke was co-fired with coal* (in large power station) (Izquierdo et al., 2007), increasing petcoke fraction added to coal increased the leachable content of V, not only absolutely (7 mg/kg and 190 mg/kg V in the leachables) but also relatively (1.8% from 380 mg/kg V in FA1 and 4.9% from 3890 mg/kg V in FA2). As it cannot be fully attributed to pH change (pH1 = 8.1 and pH2 = 8.3), rather, leachable amount reflects total V levels and probably also different modes of occurrence in low- and high- petcoke FAs (Izquierdo et al., 2007). The authors concluded that higher proportion of organically bound V (in 24% petcoke blend) somewhat hinders its assimilation to aluminosilicates thereby increasing its leachability in water. It is worth mentioning in this context that V concentration in the leachate

from high-petcoke FA is quite high (190 mg/kg) and might be potentially harmful (Izquierdo et al., 2007); therefore, extraction of V from such FA would be practical thereby enhancing its utilization options/possibilities.

6.2. Effect of pH

During the leaching, one of the most crucial affecting factors is pH; its effect on dissolved V concentration in the leachate is in some instances even more important than total V concentration in ash/slag (Silva et al., 2010). Literature results related to the effect of pH on V leachability from coal combustion ash and slag wastes are listed in Table 4.

The data summarized in Table 4 document that pH of the solution plays an important role in elemental mobility, which was in most studies the highest at pH \approx 2 (Zhao et al., 2020; Zhang et al., 2019b; Stefaniak et al., 2018; Sandeep et al., 2016). In such studies, pH of the solution was adjusted by adding HNO₃ (or NaOH). In the review paper by Izquierdo and Querol (2012), maximum V mobility was at pH= 7–11 (from evaluated range of 4–14), which corresponds with moderately higher (local maximum) leaching of V in pH= 7–9 in the study of (Zhao et al. (2020)).

Other studies evaluate V mobility in dependence of natural pH of the suspension ash/slag–water without any pH adjustment (Lieberman et al., 2020; Silva et al., 2010). In case of high-S and -Ca FAs, V mobility (in 24-h experiments) decreased with pH increase from 9.98 to 12.17 (Lieberman et al., 2020; Silva et al., 2010), which agrees with the conclusion of Nugteren et al. (2002) that due to Ca extraction, pH increases with time in longer experiments and it might result in re-precipitation of elements that have already been extracted at earlier stages. According to Izquierdo and Querol (2012), in higher-Ca systems, the formation of calcium vanadates is an important factor in terms of V solubility. This might be one of the reasons why some authors (Nugteren

Table 4

The effect of pH on leachability of V from ash and slag wastes.

Combustion	Leaching	results	Ref.
5 PCC FAs (4 ESP, 1 FF), sole coal	US EPA LEAF 1313 test, pH≈ 2–13 (HNO ₃ -NaOH), L/S= 10	Maximum V solubility at pH \approx 2, high also at pH \approx 7–9, the lowest at pH \approx 4 or 11–12	(Zhao et al., 2020)
1FA (FF) + 1BA, CFB, alkaline coal	US EPA LEAF 1313 test, $pH\approx 2-13$ (HNO ₃ -NaOH), L/S=10	Maximum V solubility at pH≈ 2 (FA and BA)	(Zhang et al., 2019b)
PCC co-firing with off gas fuel, class F FA of sialic type	US EPA LEAF 1313 test, $pH\approx 2-13$ (HNO ₃ -NaOH), L/S= 10	Maximum V solubility at $pH \approx 2$ (FA from sole coal and from co-firing with off gas fuel)	(Stefaniak et al., 2018)
2 low-grade coal FAs, Indian power plants	pH $\approx 2-12$ (HNO ₃ -NaOH), L/S= 20	Maximum V solubility at pH≈ 2 (both FAs)	(Sandeep et al., 2016)
7 high S and Ca coal FAs	EU standard Batch Leaching Test EN 12457–2, L/S= 10 L/kg, 24 h, no pH adjustment)	Solubility of V decreases as pH increases from 9.98 to 12.27 (max at 9.98)	(Lieberman et al., 2020)
7 PCC units of the same power plant (7 ESP FAs, 5 normal + 2 oil co- firing)	EU standard Batch Leaching Test EN 12457–2, L/S= 10 L/kg, 24 h, no pH adjustment	i)5 FAs from sole coal (less V in FAs): pH 9.9–10.4, higher V mobility ii)2FAs from oil co- firing (more V in FAs): pH 5.3–6.2, lower V mobility	(Silva et al., 2010)

et al., 2002) recommend incorporation of a prewashing step with water prior to using the extraction agents – it leads to higher elements removal rates and/or lower reagent consumption; even prior to bioleaching procedures (Jadhav and Hocheng, 2015).

Natural pH of the leachate is affected by coal characteristics and by co-combustion of other materials with coal. Comparison of PCC FAs from oil co-firing with normal coal combustion FAs (collected at the same combustion unit) documents diametrically different V behaviour (Silva et al., 2010). Normal coal combustion FAs with low V levels provided alkaline (pH = ca. 10) leachate with higher V concentration in comparison with pH = ca. 6 and higher V concentrations in the leachates from oil co-firing FAs (Silva et al., 2010). These results were interpreted through different modes of occurrence of V in FAs/leachates during normal combustion and co-firing with oil because element in oxyanionic form is expected to be mobile under neutral and alkaline conditions while cationic species are mobile in acidic environments (Silva et al., 2010; Cornelis et al., 2008).

6.3. Acid leaching

Various acids can be used for V leaching from ash and slag wastes (e. g., HCl or HNO_3) (Wang et al., 2004) but absolutely the most widely used is diluted H_2SO_4 .

6.3.1. Leaching by diluted H_2SO_4

Literature results on H_2SO_4 leaching of coal FAs (and other related materials) are listed in Table 5. Many researchers use increased temperature for H_2SO_4 leaching, such as 60–80 °C (Nazari et al., 2014; Chen et al., 2010) or 90 °C (Nazari et al., 2014; Chen et al., 2010); leachability of V can be improved also by using ultrasound (Masoum et al., 2021) or

Table 5

Leached percentages of V (and other elements) during H₂SO₄ leaching of FAs.

Sample	Procedure	Leached percentages	Ref.
Coal FA	1.5 M H ₂ SO ₄ , 25 °C, 3 h 500 g FA / 1 L leachant	21% V, 2.4% Ga, 60% B, 12% Cr, 25% Mn	(Tsuboi et al., 1991)
3 lignite FAs	H_2SO_4 , pH \approx 1, 25 °C, 14 days	57.4%, 72% and 86.6% V, ca. 80% Cd, Zn, U, Th, ca. 60% Cu, Co, Ni	(Seferinoglu, 2003)
 i) 2 lignite ashes (294 and 265 ppm V) ii) 1 wood ash (6 ppm V) iii) 1 asphaltite ash (3990 ppm V) 	1 M H_2SO_4, S: L= 1:10, 10 days, pH ≈ 1	 i) 60% and 95% ii) ≈ 100% iii) below 80% 	(Paul et al., 2006)
Power plant FA	i) 25% H ₂ SO ₄ , S: L= 7.5, 62.5 °C, 3 h ii) 19.47% H ₂ SO ₄ , S: L= 9.15, 80 °C, 2 h	i) 99.89% V 62.56% Ni ii) 94.28% V 81.01% Ni	(Nazari et al., 2014)
Coal FA (50 ppm V, 11 ppm Y)	32.18% H ₂ SO ₄ , 15.67% H ₂ O ₂ , 90 °C, 1 h, S/L= 0.06%, ultrasonic bath	100% V 97% Y	(Masoum et al., 2021)
Stone coal (1.09% V ₂ O ₅)	i) 6 M H ₂ SO ₄ , S: L= 1:3, 80 °C, 2 h ii) 6 M H ₂ SO ₄ , S: L= 1:3, 90 °C, 3–5 h, 6% helping reagent	i) 79.88% V ii) 95.86% V	(Chen et al., 2010)
i) 3 Heavy oil FAs (1.3–3.3% V) ii) 1 Orimulsion FA (11.7% V)	i) 2 M H ₂ SO ₄ , boiling temperature, L/S= 3 mL/g ii) 2 M (3 M) H ₂ SO ₄ , boiling temperature, L/S= 3 mL/g	i) 82%, 90% and 98% Va ii) 80% (95%) Va	(Vitolo et al., 2000)

^a rough estimation – data from graph

adding other reagents (e.g., H_2O_2) (Masoum et al., 2021; Chen et al., 2010). Even if extraction yields are quite variable (due to different ash/slag characteristics and the procedures used), in most cases it exceeds 95%. However, the extraction percentage should be evaluated in the context of bulk V content in the leached material; for example, 80% V leached from 3990 ppm in asphaltite ash provide more V than 100% recovery from wood ash with 6 ppm V (Paul et al., 2006).

Ashes provide better V extraction yields (when leached by H_2SO_4) than their coal counterparts (Seferinoglu, 2003) which is consistent with better V dissolution in 6% HCl and 6% HNO₃ out of ashes than original coals (Wang et al., 2004). It can be attributed to breakage of stable chelates of V porphyrins present in original fuel during the combustion (Paul et al., 2006).

It is practical from economic point of view if extraction procedure is slightly modified (thereby losing in some cases a few % in V yield) in order to extract within the same procedure also other elements, such as Y (Masoum et al., 2021) or Ni (Nazari et al., 2014). And vice versa – V can be "co-leached" during the H₂SO₄ extraction of other elements, such as Ce or Ga; then the retention efficiency is naturally somewhat lower (Yermagambet et al., 2018).

However, radioactive elements are also leached out of FA if treated with $H_2SO_4 - 87\%$ and 86% recovery rates were obtained for U and Th (Yermagambet et al., 2018).

Studies evaluating "co-extraction" of V and other elements during the H_2SO_4 leaching of ashes/slags are rather scarce. Nevertheless, H_2SO_4 is commonly used for the extraction of other elements in studies not including V (Table 6), such as Fe (Seidel and Zimmels, 1998), Al (Wei et al., 2018; Seidel and Zimmels, 1998; Valeev et al., 2022; Matjie and Bunt, 2005), As (Kashiwakura et al., 2010; Bradlo et al., 2015), Se (Kashiwakura et al., 2011), Y and Mn (Bradlo et al., 2015) or REE (Kashiwakura et al., 2013; Peiravi et al., 2017; Perämäki, 2014), as is shown in Table 6.

Even if in the aforementioned studies V was not evaluated, as the conditions are quite similar to those used for V leaching (Table 5), it can be expected that major proportion of V has been dissolved in these studies as well. On the one hand, it is advantageous if within one extraction step (using one leachant) multiple valuable elements are dissolved; on the other hand, low selectivity during the leaching makes higher demands on further processing of the leachate during the separation of the individual elements.

Since amorphous Al-bearing phase and Si-Al glass in coal ash are common host of V (and other trace elements) (Yang et al., 2020; Font et al., 2007), procedures improving Al leachability were tested also for the dissolution of V (and other trace elements) (Yang et al., 2020). However, even if thermal treatment of coal ash (originated from the combustion at 900 °C) with (NH₄)₂SO₄ at 400 °C (1 h) enhanced the Al solubility, the mobility of V (related to ash without (NH₄)₂SO₄ treatment) was significantly lower due to V co-precipitation with other ions. Namely at the temperatures higher than 60 °C, the formation of Fe- and Ti-bearing hydrate resulted in a significant decrease in V leaching efficiency (Yang et al., 2020). However, such coprecipitated V (and also Ga) ions can be quite easily re-dissolved by 0.1 M H₂SO₄ (Yang et al., 2020; Font et al., 2007). Then, the dissolved V was higher than in case of coal ash without (NH₄)₂SO₄ treatment (Yang et al., 2020; Font et al., 2007).

6.3.2. Leaching by diluted HCl

Dissolution of V in diluted HCl is generally lower than that in diluted H₂SO₄; nevertheless, 6% HCl (95 °C, 1 h) dissolved ca. 60–70% V from 3 coal ashes prepared in laboratory at 550–1100 °C (Wang et al., 2004). It was shown that Ni, Co, Cr, and Mn can be dissolved as well, but mobilities of namely Ni and Co were strongly dependent on ashing temperature with maximum dissolution observed after ashing at ca. 550 °C (and declining with increasing ashing temperature up to 1100 °C). Similar behaviour was also observed for Cr (or Mn) but to a lesser extent. At ca. 850 °C (simulating FBC), mobilities of Ni, Co, and Cr were ca. 50%, leachabilities of Mn were higher (ca. 80%) and not so temperature

Table 6

Results of studies on H₂SO₄ leaching of FAs not including V.

Sample	Procedure	Leached	Ref.
High-Al coal FA	H_2SO_4 :FA = 1.2	86% alumina	(Wei et al., 2018)
	Conc. H ₂ SO ₄		
	t > 230 °C		
	100 min		
Coal FA	Dilute H ₂ SO ₄ (1 or 0.1 M)	Rapid dissolution of As, decrease in solubility with increasing pH	(Kashiwakura et al.,
	10 min		2010)
	100 g FA – 1 L H ₂ SO ₄		
	Laboratory temperature		
ESP FA (bitum.	рН 1–13	At pH = 1, high dissolution of elements forming oxygen anions (As, Cr, B, Mo, Sb,	(Bradlo et al., 2015)
coal+biomass)	(H ₂ SO ₄ -NaOH)	Se, V, and W)	
Coal FA	5 g FA: 500 mL H ₂ SO ₄ , 30–80 °C, 2 h	45% dissolution rate of La, similar behaviour for Ce, Sc, Y, Pr, Nd, Sm, Gd, Eu, Tb,	(Kashiwakura et al.,
		Dy, Ho, Er, Tu, Yb, Lu	2013)
Coal ash	H ₂ SO ₄ , 30–80 °C, 2 h	8–45% REE	(Peiravi et al., 2017)
Coal FA	100 g FA – 1 L H ₂ SO ₄ (0.1 or 1 M),	Rapid dissolution of Se, decrease in solubility with increasing pH	(Kashiwakura et al.,
	25 °C, 600 s		2011)
Industrial FA	0.4 M H ₂ SO ₄	70% of the REE	(Perämäki, 2014)
	Ultrasonic bath after oxalic acid pre-		
	treatment		
	18 min		

dependent. These values (related to ashes from 3 different coals) are mutually similar and might have been used as rough estimation of behaviour of FBC ash in terms of the temperature; nevertheless, during FBC huge amount of calcareous additives are typically used, which could affect V leaching behaviour due to formation of Ca-V compounds (Nugteren et al., 2002; Izquierdo and Querol, 2012). Industrial coal FA was leached by cold diluted HCl (0.5 M) as well providing 63% V recovery during 24 h and 67% after 7 days (Harris and Silberman, 1983).

Other studies using HCl do not include V. Using 1 M HCl (85 °C, 4 h) ca. 80–90% REE was dissolved from PCC FAs (Middleton et al., 2020) or using cold HCl (HCl: H₂O =2:1, vol/vol) about 50% of Y was dissolved out of FBC FAs (Bartoňová et al., 2018). Dilute HCl was also tested for Ga recovery from coal FA – Fang and Gesser (1996) used 2 M HCl (24 h, ambient temperature) while Gutierrez (1997) used 6 M HCl (1 h, boiling point). Hence, in case of diluted HCl (temperature up to ca. 95 °C), V could be partially leached along with other elements, such as Ni, Co, Cr, Mn, Y, or REE. Compared to H₂SO₄, extraction yields are generally lower, but it can be used at least as an efficient "cleaning application" facilitating further use of the leached residues.

6.4. Alkaline leaching/roasting

Literature on alkaline leaching of coal combustion wastes and related V data are rather scarce. Using weak base as a leaching agent (0.5 M NH₄OH) dissolved during 24 h 29 ppm out of total 260 ppm V in coal FA, which is ca. 11% (Harris and Silberman, 1983). Font et al. (2007) reported < 64% V extraction yields from coal IGCC FA when using extreme alkaline extraction conditions (the recovery of Ga was much better – 98.8%).

6.4.1. Comparison of H₂SO₄ and NaOH leaching

Efficiency of alkaline and acidic leaching of V in ashes from soft asphalt fuels (15% V) was compared by Chmielewski et al. (1997) rejecting acid leaching as ineffective. Using alkaline leaching by 30% NaOH (2 h, 100–110 °C), the leaching efficiency of V was 94% and can be described by Eqs. (3) and (4) (Chmielewski et al., 1997):

$$2 \operatorname{VO}_2^+ + 4 \operatorname{NaOH} \rightarrow \operatorname{Na}_4 \operatorname{V}_2 \operatorname{O}_7 + \operatorname{H}_2 \operatorname{O} + \operatorname{H}^+$$
(3)

$$VO_2^+ + 3 \text{ NaOH} \rightarrow Na_3VO_4 + H_2O + H^+$$
(4)

After oxidation of V(4 +) to V(5 +) by 30% H_2O_2 , neutralization of alkaline leachate to ca. pH= 8 by H_2SO_4 and adding ammonium sulphate (or chloride), ammonium metavanadate precipitate is formed (Eq. (5)) (Chmielewski et al., 1997):

 $Na_4V_2O_7 + 4NH_4^+ \rightarrow 2NH_4VO_3 + 2NH_3 + 4Na^+ + H_2O$ (5)

Navarro et al. (2007) compared 0.5 M H₂SO₄ and 2 M NaOH leaching of oil FA (1.6% V) followed by the second step of V recovery from leachates by solvent extraction and selective precipitation. Both procedures led to ca. 98% recovery; however, using 2 M NaOH required more leaching repetitions (extraction steps). Despite this, alkaline leaching was evaluated to be better due to its selectivity for V (vs. other transition metals). Precipitation was preferred to amine extraction due to better selectivity as well.

Comparison of acid and alkaline leaching of dumped FA from crude oil power plant by 0–5 M NaOH and 0–5 M H₂SO₄ (96 h) exhibited comparable results (being for H₂SO₄ slightly better). Both leaching agents were also evaluated in terms of the second step that followed – V retention from leachate on chitosan or V extraction with amine extractants – for both these procedures, acidic solution was evaluated as more advantageous. Therefore, extraction by H₂SO₄ was recommended for this purpose (Guibal et al., 2003).

6.4.2. Alkaline roasting - leaching method

If V occurs in the form of V-bearing spinels $(Fe,Mg,Mn)_x(V,Cr)_{3-x}O_4$, alkaline roasting typically used for V slags may be efficient procedure (Li et al., 2016, 2015). As roasting with some sodium salts often generates corrosion gases (HCl, Cl₂ etc.), calcification roasting may be a cleaner procedure converting V-bearing spinels by CaO or CaCO₃ to Ca-vanadates, according to Eqs. (6) and (7) (Li et al., 2016):

 $4 \text{ CaO} + 4 \text{ FeVO}_4 + 5 \text{ O}_2 \rightarrow 4 \text{ Ca(VO}_3)_2 + 2 \text{ Fe}_2\text{O}_3$ (6)

$$Ca(VO_3)_2 + CaO \rightarrow Ca_2V_2O_7 \tag{7}$$

Which is followed by H_2SO_4 leaching (Eqs. (8) and (9)) (Li et al., 2016):

$$Ca(VO_3)_2 + 2 H_2SO_4 \rightarrow (VO_2)_2SO_4 + CaSO_4 + 2 H_2O$$
 (8)

$$Ca_2V_2O_7 + 3 H_2SO_4 \rightarrow (VO_2)_2SO_4 + 2CaSO_4 + 3 H_2O$$
 (9)

V retention in the aforementioned experiments were 96% and 98.9%, which documents efficiency of this approach.

6.5. Bioleaching

Not only chemical leaching but also biotechnological approaches of metal recovery from waste materials have been tested. This approach is based on microbial activities converting insoluble metal compounds into water-leachable species. Advantages of bacterial leaching over the other methods are lower cost, using normal conditions, low energy requirements and small areas for installation. Disadvantages are low growth rate of bacteria, low biooxidation rates and low selectivity (Snegirev and Sabirova, 2021; Park and Liang, 2019; Meer and Nazir, 2018).

Particular attention has been paid to using autotrophic bacteria (i.e. bacteria that do not require carbon substrate and use energy from oxidation of S-, Fe- or N-species) (Snegirev and Sabirova, 2021; Park and Liang, 2019; Meer and Nazir, 2018). In case of sulphidic materials, the most common is using chemolithotrophic bacteria, such as Acidothiobacillus sp. that convert metal sulphides into metal sulphates generating H₂SO₄ (acidolysis) (Lee and Pandey, 2012; Brombacher et al., 1997). In case of non-sulfidic materials (minerals, wastes etc.), heterotrophic microorganisms (which require a substrate with organic carbon) are widely used. As a product of the metabolism of this substrate, such microorganisms produce acidic compounds, e.g. acetic, citric, oxalic or gluconic acids, able to dissolve oxides, silicates, carbonates, hydroxides etc. by reduction acidolysis and/or complexation (Jain and Sharma, 2004). Heterotrophic microorganisms used for bioleaching include bacteria (Bacillus pseudomonas etc.) (Jain and Sharma, 2004) or fungi (Aspergillus, Penicillinum etc.) (Jain and Sharma, 2004; Lee and Pandey, 2012). Heterotrophic bacteria and fungi can tolerate high pH (Mirazimi et al., 2015).

Traditionally, if bioleaching was applied on coal combustion ashes/ slags, it was mostly conducted with the aim to extract Al by Aspergillus niger (Singer et al., 1982; Torma and Singh, 1993) or Thiobacillus thiooxidans (Fass et al., 1994; Seidel and Zimmels, 1998), Fe by Thiobacillus thiooxidans (Seidel and Zimmels, 1998) or Ti by Rhodococcus sp (Shabtai et al., 1993). Results related to V bioleaching from coal ashes /slags are quite scarce, so some other wastes are mentioned herein as well. Bosecker (1987) concluded that if Thiobacillus thiooxidans was applied on 6 MSWI FAs, the V extraction recovery during bacterial leaching was as effective as in case of chemical leaching using H₂SO₄ (in case of 4 samples out of 6, the mobilities were almost 100%). In case of LD converter slag with 1.97% V2O5, Mirazimi et al. (2015) observed better V removal by bacteria (ca. 80%) than by fungi (ca. 70%). In this study autotropic (Acidothiobacillus thiooxidans) and heterotrophic (Pseudomonas putida) bacteria and heterotrophic fungi (Aspergillus niger) were used. Rastegar et al. (2015) achieved 82% V mobility by Acidithiobacilus ferrooxidans if applied on V-rich oil-fired ash with 49 g/kg V. Moreover, this procedure also mobilized 86% Ni and 87% Cu. A novel strain (Bacillus mucilaginosus) was tested for the V extraction from low-grade stone coal. The results from direct bioleaching with 57% recovery was significantly improved by the calcination of the stone coal prior to bioleaching achieving 85% V mobility (Dong et al., 2020).

Recently, promising results of V bioleaching from coal ash/slag have appeared.

Chronologically, Jadhav and Hocheng (2015) extracted V from pre-washed thermal power plant FA by citric acid producing *Aspergillus niger* 34770 Culture (30 °C, 4 h, S/L=1 g/100 mL) with extraction recovery of ca. 100%. During the same procedure, almost 100% recovery was obtained also for Cd, Co, Se, Ca, Cu, Mn, Fe, Al, Zn, Ti, B, K, and Mg, while in case of Cr, Ni, As, and Pb, it was slightly lower (93%, 83%, 78% and 70%, respectively). After the bioleaching, the seed germination, the root and shoot growth of *V. radiata* was substantially improved (in comparison with the original unleached FA).

Gluconic acid producing *Acidomonas methanolica* used by Kermer et al. (2017) for mobilization of elements from stabilized lignite ash provided 50–80% extraction not only for V, but also for Ce, Sr, Ti, and Zr.

Park and Liang (2019) compared bioleaching of V from coal FA (104 ppm V) by 3 microbial strains: *Candida bombicola, Phanerochaete chrysosporium* and *Cryptococcus curvatus* with extraction efficiency being 70.2%, 65.2% and 61.7%. In this study, ca. 30 elements have been evaluated and only As and Mo (and perhaps Sb) provided higher extraction recovery (> 60%) by *Phanerochaete chrysosporium* and *Cryptococcus Curvatus*. Using *Candida bombicola*, > 60% V, As, Mo, Sb, U, Sc,

Y, Dy, Er, and Yb has been leached.

The results indicate that the bioleaching efficiency is comparable to that of traditional chemical methods. For example, heterotrophic microorganisms produce organic acids (acetic, citric, oxalic, gluconic etc.) and metal complexation is common mechanism related to metal dissolution (Jain and Sharma, 2004). Using synthetic solutions of 0.01 M chelating agents for V dissolution from (260 ppm) coal ESP FA revealed 41% V dissolved by EDTA and 44% dissolved in citric acid after 7-day treatment (V recovery when using histidine or glycine was quite low - < 2%) (Harris and Silberman, 1983).

Latest papers document modern promising trends but unfortunately it does not include V data. A sequential integration approach of hydrothermal alkali treatment and bioleaching was applied to coal FA providing maximum leaching rates for Ti, Ga, Sr, Zr, Ba, and Bi being 97.6%, 71.4%, 72.4%, 87.0%, 45.6%, and 60.9%, respectively (Su et al., 2020).

7. Concluding remarks and future research perspectives

The study reviews crucial aspects related to V present in coal and corresponding ash and slag residues with particular attention paid to its potential extraction. The paper evaluates the results from traditional coal combustion as well as from modern technologies with great environmental benefits, such as co-combustion of coal and wastes or oxy-fuel combustion. The research areas evaluated in this study are:

- V levels in input materials coal, wastes, or combustion additives (including speciation and modes of occurrence)
- ii) Thermodynamic equilibrium calculations of V behaviour during the combustion
- iii) V levels in corresponding ash and slag counterparts (and distribution among bottom ash/slag and fly ash, particulate control devise rows and V speciation)
- iv) Physical separation methods particle-size separation, magnetic separation, density separation, electrostatic separation etc.
- v) Chemical extraction procedures leaching by water, various acids, or alkaline solutions including the effect of pH and roasting
- vi) Bioleaching and the comparison with traditional chemical dissolution

Average V concentrations in brown-coals and hard-coal are below 50 ppm, in brown-coal and hard-coal ashes the average levels are 100-200 ppm. If coal is co-combusted with other materials, V content in ashes/slags can be lower (e.g., in case of agricultural or forest residues) or substantially higher (namely in case of petcoke). Using desulphurization additives generally results in a "diluting" effect on V in ashes due to low V levels in most calcareous additives. Regarding the selection of optimal material serving as a source of V, high V concentrations are in residues from co-combustion of coal and petcoke. From conventional combustion residues, FAs contain typically more V than bottom ashes/ slags; FAs from the last sections (hoppers) of ESP generally contain more V that the FAs from the first and second sections but relative abundance of FAs from the last rows is quite low. Coal ashes are mostly better V source than original coals; at least in the case that the combustion temperature is not too high. Hence, if fuel is energetically used, due to quite low V volatility its fraction in the form of gaseous emissions is low (a few %) and as a result of the thermal treatment V concentrations as well as its leachability is elevated (related to original coal).

Numerous *physical separation methods* have been tested to elevate V concentrations in ashes/slags, such as particle-size fractionation, magnetic separation, density (sink-float) separation, electrostatic separation etc. None of these techniques generally cannot provide material with sufficiently high V concentration. Nevertheless, they can be used as preconcentration procedures providing fractions with increased V content (toxicity of fractions depleted in V and other toxic elements is thereby reduced facilitating their handling, utilization or meeting the threshold

limits).

Hence, if production of V is the main objective, chemical or biological approach should generally follow the physical-separation pretreatment. Effective V extraction (leaching) procedures exist that mostly achieve > 90–95% V extraction yield. The residual amount of V (a few %) left after the extraction in the ash should not pose a problem in terms of toxic V-containing leachate as it is expected to be firmly bound in the leached ash/slag residue.

Despite high leaching efficiency, chemical extraction of V from conventional ash and slag wastes is not expected to be economically attractive. Therefore, an effort paid to incorporation of V recovery into more complex extraction scheme (providing more valuable components) seems to be more promising way how to economically obtain V from common combustion wastes. For example, good V extraction yields have been reported using H₂SO₄. Individual optimization of the extraction procedure facilitates achieving good V extraction results if applied on various materials - on conventional coal ashes/slags, biomass ash, stone coal, asphaltite ash, heavy oil FAs or orimulsion FA. But only a few studies evaluate "co-extraction" of V and other elements by H₂SO₄, such as V and Y, V and Ni or V, Ce and Ge. Nevertheless, as H₂SO₄ is widely used for the extraction of various other elements (than V), such as Al, As, Se, Y, and Mn or REE, due to similarity of conditions it can be expected that major fraction of V has been dissolved in these studies as well. However, dissolution of multiple elements from FA by the same leachant makes higher demands on further processing of the leached solution during the separation of the individual elements.

V bioleaching studies are even more scarce. A few papers appeared recently document that bioleaching is another promising approach showing that not only V but also other elements could be removed from coal ash. Nevertheless, only a few microbial strains have been tested for this purpose – *Aspergillus niger, Acidomonas methanolica, Candida bombicola, Phanerochaete chrysosporium* and *Cryptococcus curvatus*. Latest research documents that efficiency of bioleaching by *Aspergillus niger* from coal FA can be substantially increased by hydrothermal alkali pretreatment step, but this study does not evaluate V (it was focused to Ti, Ga, Sr, Zr, Ba, and Bi). Almost complete removal of V, Cd, Co, Se, Ca, Cu, Mn, Fe, Al, Zn, Ti, B, K, and Mg from FA was described in literature thereby effectively reducing strong harmful effect of untreated FA on seed germination, the root and shoot growth of *V. radiata* by *Aspergillus niger* bioleaching. Hence, the "cleaned" ash with reduced toxicity is another valuable benefit.

Therefore, overall, more valuable components could be extracted out of the ash/slag at the same time (such as V, Y, Sc, Ga, REE, etc.); moreover, the "cleaned" ash/slag residues with reduced toxicity and enhanced utilization perspectives should provide other benefits (reduced cost of landfilling, economic benefit from replacement of natural resources used e.g. in civil engineering, solving the supply risk of other critical materials etc.). Thus, techno-economic studies including such overall evaluation conducted with respect to all these aspects and benefits would be undoubtedly useful.

Hitherto the research in this area was focused predominantly to conventional coal combustion residues (ashes/slags etc.). As power production is one of a major CO_2 -emitting sector, there is an intensive decarbonization effort leading to a rapid development of modern technologies mitigating CO_2 emissions, such as oxy-fuel combustion, IGCC (combined optimally with CO_2 capture), co-firing with low C or C-neutral fuels or carbon capture technologies. Hence it can be expected that new industrial wastes will be generated in near future in vast quantities, which is also worth research attention as the literature in this field is currently rather limited.

CRediT authorship contribution statement

Lucie Bartoňová: Conceptualization, Writing – original draft, Writing – review & editing. **Helena Raclavská**: Writing – original draft, Writing – review & editing, Supervision. **Jan Najser**: Writing – original draft.

Declaration of Competing Interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Acknowledgement

This paper was supported by the research project of the Ministry of Education, Youth and Sport of the Czech Republic: "OP RDE grant number CZ.02.1.01/0.0/0.0/16_019/0000753 Research Centre for Low-carbon Energy Technologies".

References

- Alastuey, A., Jiménez, A., Plana, F., Querol, X., Suárez-Ruiz, I., 2001. Goechemistry, mineralogy, and technological properties of the main Stephanian (Carboniferous) coal seams from Puertollano Basin, Spain. Int. J. Coal Geol. 45, 247–265.
- Bacci, P., Del Monte, M., Longhetto, A., Piano, A., Prodi, F., Redaelli, P., Sabbioni, C., Ventura, A., 1983. Characterization of the particulate emission by a large oil fuel fired power plant. J. Aerosol Sci. vol. 14, 557–572. https://doi.org/10.1016/0021-8502(83)90011-3.
- Badenhorst, C., Santos, C., Lázaro-Martínez, J., Białecka, B., Cruceru, M., Guedes, A., Guimarães, R., Moreira, K., Predeanu, G., Suárez-Ruíz, I., Cameán, I., Valentim, B., Wagner, N., 2020. Assessment of graphitized coal ash char concentrates as a potential synthetic graphite source. Minerals vol. 10. https://doi.org/10.3390/ min10110986.
- Bai, X., Li, W., Chen, Y., Jiang, Y., 2007. The general distributions of trace elements in Chinese coals: (in Chinese with English abstract). Coal Qual. Technol. 1, 1–4.
- Barceloux, D., Barceloux, D., 1999. Vanadium. J. Toxicol.: Clin. Toxicol. vol. 37, 265–278. https://doi.org/10.1081/CLT-100102425.
- Bartoňová, L., 2015. Unburned carbon from coal combustion ash: An overview. Fuel Process. Technol. vol. 134, 136–158. https://doi.org/10.1016/j.fuproc.2015.01.028.
- Bartoňová, L., Klika, Z., 2014. Effect of CaO on retention of S, Cl, Br, As, Mn, V, Cr, Ni, Cu, Zn, W and Pb in bottom ashes from fluidized-bed coal combustion power station. J. Environ. Sci. vol. 26, 1429–1436. https://doi.org/10.1016/j.jes.2014.05.008.
- Bartoňová, L., Raclavská, H., 2022. Behavior of Cr during coal combustion: An overview. Fuel vol. 322. https://doi.org/10.1016/j.fuel.2022.124210.
- Bartoňová, L., Kořistková, M., Kolat, P., Klika, Z., Čech, B., 2010. Study of unburned carbon and volatility of elements during co-combustion of coal and waste alternative fuel, Chemical and Process Engineering, 31, 725–739.
- Bartoňová, L., Serenčíšová, J., Čech, B., 2018. Yttrium partitioning and associations in coal-combustion ashes prior to and after their leaching in HCl. Fuel Process. Technol. vol. 173, 205–215. https://doi.org/10.1016/j.fuproc.2018.01.011.
- Bartoňová, L., Raclavská, H., Čech, B., Kucbel, M., 2019. Behavior of Pb During Coal Combustion: An Overview. Sustainability vol. 11. https://doi.org/10.3390/ su11216061.
- Bartoňová, L., Raclavská, H., Čech, B., Kucbel, M., 2020. Behavior of Cd during Coal Combustion: An Overview. Processes vol. 8. https://doi.org/10.3390/pr8101237.
- Bergh, J., Falcon, R., Falcon, L., 2011. Trace element concentration reduction by beneficiation of Witbank Coalfield no. 4 Seam. Fuel Process. Technol. vol. 92, 812–816. https://doi.org/10.1016/j.fuproc.2010.09.035.
- Berland, T., Pflughoeft-Hassett, D., Dockter, B., Eylands, K., Hassett, D., Heebink, L., 2003. Review of handling and use of FGD material. CARRC topical report. Energy Environ. Res. Cent., Gd. Forks, ND, USA 37 (pp).
- Bielowicz, B., 2020. Ash Characteristics and Selected Critical Elements (Ga, Sc, V) in Coal and Ash in Polish Deposits. Resources vol. 9. https://doi.org/10.3390/ resources9090115.
- Bielowicz, B., 2021. Selected critical raw materials in waste from coal gasification in Poland. Energies vol. 14. https://doi.org/10.3390/en14238071.
- Bosecker, K., 1987. Microbial recycling of mineral waste products. Acta Biotechnol. vol. 7, 487–497. https://doi.org/10.1002/abio.370070602.
- Bouška, V., Pešek, J., 1999. Quality parameters of lignite of the North Bohemian Basin in the Czech Republic in comparison with the world average lignite. Int. J. Coal Geol. vol. 40, 211–235. https://doi.org/10.1016/S0166-5162(98)00070-6.
- Bradlo, D., Zukowski, W., Czupryński, P., Witkowski, K., 2015. Potential of metal recovery from coal combustion products. Part II. Leaching of selected elements. Inz. Miner. 16, 163–170.
- Brombacher, C., Bachofen, R., Brandl, H., 1997. Biohydrometallurgical processing of solids: a patent review. Appl. Microbiol. Biotechnol. vol. 48, 577–587. https://doi. org/10.1007/s002530051099.
- Cairncross, B., Hart, R., Willis, J., 1990. Geochemistry and sedimentology of coal seams from the Permian Witbank Coalfield, South Africa; a means of identification. Int. J. Coal Geol. vol. 16, 309–325. https://doi.org/10.1016/0166-5162(90)90056-5.
- Chen, G., Sun, Y., Wang, Q., Yan, B., Cheng, Z., Ma, W., 2019. Partitioning of trace elements in coal combustion products: A comparative study of different applications in China. Fuel vol. 240, 31–39. https://doi.org/10.1016/j.fuel.2018.11.131.

Chen, X., Lan, X., Zhang, Q., Ma, H., Zhou, J., 2010. Leaching vanadium by high concentration sulfuric acid from stone coal. Trans. Nonferrous Met. Soc. China vol. 20, s123–s126. https://doi.org/10.1016/S1003-6326(10)60025-8.

W.Chinach, Powstaje Rekordowy Magazyn Energii Oparty o Nowe Ogniwa Wanadowe-Swiat0ZE.pl. https://swiatoze.pl/w-chinach-powstaje-rekordowy-magazyn-energiioparty-o-noweogniwa- wanadowe/ (accessed on 13 May 2020),, (n.d.). https:// swiatoze.pl/w-chinach-powstaje-rekordowy-magazyn-energii-oparty-onoweogniwa- wanadowe/ (accessed (accessed on 13 May 2020).

Chmielewski, A., Urbański, T., Migdał, W., 1997. Separation technologies for metals recovery from industrial wastes. Hydrometallurgy vol. 45, 333–344. https://doi.org/ 10.1016/S0304-386X(96)00090-4.

Ciesielczuk, J., Misz-Kennan, M., Hower, J., Fabiańska, M.J., 2014. Mineralogy and geochemistry of coal wastes from the Starzykowiec coal-waste dump (Upper Silesia, Poland. Int. J. Coal Geol. vol. 127, 42–55. https://doi.org/10.1016/j. coal.2014.02.007.

Coppola, A., Scala, F., 2021. Chemical looping for combustion of solid biomass: a review. Energy Fuels vol. 35, 19248–19265. https://doi.org/10.1021/acs. energyfuels.1c02600.

Coppola, A., Scala, F., Azadi, M., 2022. Direct dry carbonation of mining and industrial wastes in a fluidized bed for offsetting carbon emissions. Processes vol. 10. https:// doi.org/10.3390/pr10030582.

Córdoba, P., Ochoa-Gonzalez, R., Font, O., Izquierdo, M., Querol, X., Leiva, C., López-Antón, M., Díaz-Somoano, M., Rosa Martinez-Tarazona, M., Fernandez, C., Tomás, A., 2012. Partitioning of trace inorganic elements in a coal-fired power plant equipped with a wet Flue Gas Desulphurisation system. Fuel vol. 92, 145–157. https://doi.org/10.1016/j.fuel.2011.07.025.

Cornelis, G., Johnson, C., Gerven, T., Vandecasteele, C., 2008. Leaching mechanisms of oxyanionic metalloid and metal species in alkaline solid wastes: A review. Appl. Geochem. vol. 23, 955–976. https://doi.org/10.1016/j.apgeochem.2008.02.001.

Cornelius, M., Ameh, A., Eze, C., Fatoba, O., Sartbaeva, A., Petrik, L., 2021. The behaviour of rare earth elements from south african coal fly ash during enrichment processes: wet, magnetic separation and zeolitisation. Minerals vol. 11. https://doi. org/10.3390/min11090950.

Corsaro, A., Raclavská, H., Hlavsová, A., Frydrych, J., Juchelková, D., 2016. Perennial grasses as prospective energy sources. Energy Sources, Part A: Recovery, Util., Environ. Eff. vol. 38, 1206–1211. https://doi.org/10.1080/15567036.2014.886000.

Cutruneo, C., Oliveira, M., Ward, C., Hower, J., de Brum, I., Sampaio, C., Kautzmann, R., Taffarel, S., Teixeira, E., Silva, L., 2014. A mineralogical and geochemical study of three Brazilian coal cleaning rejects: Demonstration of electron beam applications. Int. J. Coal Geol. vol. 130, 33–52. https://doi.org/10.1016/j.coal.2014.05.009.

Czech, T., Marchewicz, A., Sobczyk, A., Krupa, A., Jaworek, A., Śliwiński, Ł., Rosiak, D., 2020. Heavy metals partitioning in fly ashes between various stages of electrostatic precipitator after combustion of different types of coal. Process Saf. Environ. Prot. vol. 133, 18–31. https://doi.org/10.1016/j.psep.2019.10.033.

Dahl, O., Nurmesniemi, H., Pöykiö, R., Watkins, G., 2009. Comparison of the characteristics of bottom ash and fly ash from a medium-size (32 MW) municipal district heating plant incinerating forest residues and peat in a fluidized-bed boiler. Fuel Process. Technol. vol. 90, 871–878. https://doi.org/10.1016/j. fuproc. 2009.04.013.

Dai, S., Li, D., Chou, C., Zhao, L., Zhang, Y., Ren, D., Ma, Y., Sun, Y., 2008. Mineralogy and geochemistry of boehmite-rich coals: New insights from the Haerwusu Surface Mine, Jungar Coalfield, Inner Mongolia, China. Int. J. Coal Geol. vol. 74, 185–202. https://doi.org/10.1016/j.coal.2008.01.001.

https://doi.org/10.1016/j.coal.2008.01.001.
Dai, S., Ren, D., Chou, C., Finkelman, R., Seredin, V., Zhou, Y., 2012. Geochemistry of trace elements in Chinese coals: A review of abundances, genetic types, impacts on human health, and industrial utilization. Int. J. Coal Geol. vol. 94, 3–21. https://doi. org/10.1016/j.coal.2011.02.003.

Dai, S., Seredin, V., Ward, C., Jiang, J., Hower, J., Song, X., Jiang, Y., Wang, X., Gornostaeva, T., Li, X., Liu, H., Zhao, L., Zhao, C., 2014a. Composition and modes of occurrence of minerals and elements in coal combustion products derived from high-Ge coals. Int. J. Coal Geol. vol. 121, 79–97. https://doi.org/10.1016/j. coal.2013.11.004.

Dai, S., Luo, Y., Seredin, V., Ward, C., Hower, J., Zhao, L., Liu, S., Zhao, C., Tian, H., Zou, J., 2014b. Revisiting the late Permian coal from the Huayingshan, Sichuan, southwestern China: Enrichment and occurrence modes of minerals and trace elements. Int. J. Coal Geol. vol. 122, 110–128. https://doi.org/10.1016/j. coal.2013.12.016.

Davidson, R., Santos, S., 2010. Oxyfuel combustion of pulverised coal, 2010/7: IEA Clean Coal Centre report.

Demirbas, A., 2004. Combustion characteristics of different biomass fuels. Prog. Energy Combust. Sci. vol. 30, 219–230. https://doi.org/10.1016/j.pecs.2003.10.004.

Díaz-Somoano, M., Unterberger, S., Hein, K., 2006. Prediction of trace element volatility during co-combustion processes. Fuel vol. 85, 1087–1093. https://doi.org/10.1016/ j.fuel.2005.10.013.

Dong, Y., Lin, H., Liu, Y., Zhao, Y., 2020. Blank roasting and bioleaching of stone coal for vanadium recycling. J. Clean. Prod. vol. 243 https://doi.org/10.1016/j. jclepro.2019.118625.

Duan, L., Duan, Y., Sarbassov, Y., Li, Y., Anthony, E., 2015. SO3 formation under oxy-CFB combustion conditions. Int. J. Greenh. Gas. Control vol. 43, 172–178. https:// doi.org/10.1016/j.ijggc.2015.10.028.

Duchesne, M., Nakano, J., Hu, Y., MacLennan, A., Bennett, J., Nakano, A., Hughes, R., 2018. Synchrotron-based X-ray absorption spectroscopy study of vanadium redox speciation during petroleum coke combustion and gasification. Fuel vol. 227, 279–288. https://doi.org/10.1016/j.fuel.2018.04.104.

Eskenazy, G., 2009. Trace elements geochemistry of the Dobrudza coal basin, Bulgaria. Int. J. Coal Geol. vol. 78, 192–200. https://doi.org/10.1016/j.coal.2009.01.005. Eskenazy, G., Stefanova, Y., 2007. Trace elements in the Goze Delchev coal deposit, Bulgaria. Int. J. Coal Geol. vol. 72, 257–267. https://doi.org/10.1016/j. coal.2007.03.002.

European Commission, Critical Raw Materials Resilience: Charting a Path towards greater Security and Sustainability: Communication from the Commission to the European Parliament, the Council, the European Economic and Social Committee and the Committee of the Regions, (2020). (https://eur-lex.europa.eu/legal-content/ EN/TXT/?uri=CELEX:52020DC0474).

Fang, Z., Gesser, H., 1996. Recovery of gallium from coal fly ash. Hydrometallurgy vol. 41, 187–200. https://doi.org/10.1016/0304-386X(95)00055-L.

Fass, R., Geva, J., Shalita, Z., White, M., Fleming, J., 1994. Bioleaching method for the extraction of metals from coal fly ash using Thiobacillus. US Patent 5,278,069.

Faure, K., Willis, J., Claris Dreyer, J., 1996. The grootegeluk formation in the Waterberg Coalfield, South Africa: facies, palaeoenvironment and thermal history — evidence from organic and clastic matter. Int. J. Coal Geol. vol. 29, 147–186. https://doi.org/ 10.1016/0166-5162(95)00029-1.

Finkelman, R., Palmer, C., Wang, P., 2018. Quantification of the modes of occurrence of 42 elements in coal. Int. J. Coal Geol. vol. 185, 138–160. https://doi.org/10.1016/j. coal.2017.09.005.

Fleischer, M., 1953. Recent Estim. Abundances Elem. Earth's Crust.

Folgueras, M., Díaz, R., Xiberta, J., Alonso, M., 2007. Effect of Inorganic Matter on Trace Element Behavior during Combustion of Coal–Sewage Sludge Blends. Energy Fuels vol. 21, 744–755. https://doi.org/10.1021/ef060536r.

Font, O., Querol, X., Juan, R., Casado, R., Ruiz, C., López-Soler, Á., Coca, P., Peña, F., 2007. Recovery of gallium and vanadium from gasification fly ash. J. Hazard. Mater. vol. 139, 413–423. https://doi.org/10.1016/j.jhazmat.2006.02.041.

Font, O., Querol, X., Izquierdo, M., Alvarez, E., Moreno, N., Diez, S., Álvarez-Rodríguez, R., Clemente-Jul, C., Coca, P., Garcia-Peña, F., 2010. Partitioning of elements in a entrained flow IGCC plant: Influence of selected operational conditions. Fuel vol. 89, 3250–3261. https://doi.org/10.1016/j.fuel.2010.03.044.

Frandsen, F., Dam-Johansen, K., Rasmussen, P., 1994. Trace elements from combustion and gasification of coal—An equilibrium approach. Prog. Energy Combust. Sci. vol. 20, 115–138. https://doi.org/10.1016/0360-1285(94)90007-8.

Fu, B., Liu, G., Mian, M., Sun, M., Wu, D., 2019a. Characteristics and speciation of heavy metals in fly ash and FGD gypsum from Chinese coal-fired power plants. Fuel vol. 251, 593–602. https://doi.org/10.1016/j.fuel.2019.04.055.

Fu, B., Liu, G., Mian, M., Zhou, C., Sun, M., Wu, D., Liu, Y., 2019b. Co-combustion of industrial coal slurry and sewage sludge: Thermochemical and emission behavior of heavy metals. Chemosphere vol. 233, 440–451. https://doi.org/10.1016/j. chemosphere.2019.05.256.

Fu, B., Hower, J., Li, S., Huang, Y., Zhang, Y., Hu, H., Liu, H., Zhou, J., Zhang, S., Liu, J., Yao, H., 2021. The key roles of Fe-bearing minerals on arsenic capture and speciation transformation during high-As bituminous coal combustion: Experimental and theoretical investigations. J. Hazard. Mater. vol. 415 https://doi.org/10.1016/j. ihazmat.2021.125610.

Fu, B., Hower, J., Zhang, W., Luo, G., Hu, H., Yao, H., 2022. A review of rare earth elements and yttrium in coal ash: Content, modes of occurrences, combustion behavior, and extraction methods. Prog. Energy Combust. Sci. vol. 88 https://doi. org/10.1016/j.pecs.2021.100954.

Galetz, X., 2015, Vanadium induced corrosion in the partial oxidation (POX) process, in: NACE - International Corrosion Conference Series.

Gentzis, T., Goodarzi, F., 1999. Chemical fractionation of trace elements in coal and coal ash. Energy Sources vol. 21, 233–256. https://doi.org/10.1080/ 00908319950014867.

George, A., Dugwell, D., Kandiyoti, R., 2008. Leaching hierarchies in co-combustion residues. Energy Fuels vol. 22, 1641–1649. https://doi.org/10.1021/ef700560e.

George, A., Larrion, M., Dugwell, D., Fennell, P., Kandiyoti, R., 2010. Co-firing of Single, Binary, and Ternary Fuel Blends: Comparing Synergies within Trace Element Partitioning Arrived at by Thermodynamic Equilibrium Modeling and Experimental Measurements. Energy Fuels vol. 24, 2918–2923. https://doi.org/10.1021/ ef100001h

Glushkov, D., Shabardin, D., Strizhak, P., Vershinina, K., 2016. Influence of organic coalwater fuel composition on the characteristics of sustainable droplet ignition. Fuel Process. Technol. vol. 143, 60–68. https://doi.org/10.1016/j.fuproc.2015.11.014.

González, A., Navia, R., Moreno, N., 2009. Fly ashes from coal and petroleum coke combustion: current and innovative potential applications. Waste Manag. Res.: J. a Sustain. Circ. Econ. vol. 27, 976–987. https://doi.org/10.1177/ 0734242X09103190.

Goodarzi, F., 2002. Mineralogy, elemental composition and modes of occurrence of elements in Canadian feed-coals. Fuel vol. 81, 1199–1213. https://doi.org/10.1016/ S0016-2361(02)00023-6.

Goodarzi, F., Sanei, H., Stasiuk, L., Bagheri-Sadeghi, H., Reyes, J., 2006. A preliminary study of mineralogy and geochemistry of four coal samples from northern Iran. Int. J. Coal Geol. vol. 65, 35–50. https://doi.org/10.1016/j.coal.2005.04.014.

Guibal, E., Guzman, J., Navarro, R., Revilla, J., 2003. Vanadium Extraction from Fly Ash—Preliminary Study of Leaching, Solvent Extraction, and Sorption on Chitosan. Sep. Sci. Technol. vol. 38, 2881–2899. https://doi.org/10.1081/SS-120022577.

Guo, C., Zou, J., Ma, S., Yang, J., Wang, K., 2019. Alumina Extraction from Coal Fly Ash via Low-Temperature Potassium Bisulfate Calcination. Minerals vol. 9. https://doi. org/10.3390/min9100585.

Guo, F., Zhong, Z., 2018. Co-combustion of anthracite coal and wood pellets: Thermodynamic analysis, combustion efficiency, pollutant emissions and ash slagging. Environ. Pollut. vol. 239, 21–29. https://doi.org/10.1016/j. envpol.2018.04.004.

- Gutierrez, B., 1997. Recovery of gallium from coal fly ash by a dual reactive extraction process. Waste Manag. Res. vol. 15, 371–382. https://doi.org/10.1006/ wmre.1996.0093.
- Harris, W., Silberman, D., 1983. Time-dependent leaching of coal fly ash by chelating agents. Environ. Sci. Technol. vol. 17, 139–145. https://doi.org/10.1021/ es00109a004.
- Henke, K., 2005. Trace Element Chemistry of Fly Ashes from Co-combusted Petroleum Coke and Coal. In: World of Coal Ash (WOCA), April 11-15, 2005. Lexington, Kentucky, USA.
- Hower, J., Robertson, J., Thomas, G., Wong, A., Schram, W., Graham, U., Rathbone, R., Robl, T., 1996. Characterization of fly ash from Kentucky power plants. Fuel vol. 75, 403–411. https://doi.org/10.1016/0016-2361(95)00278-2.
- Hower, J., Trimble, A., Eble, C., Palmer, C., Kolker, A., 1999. Characterization of Fly Ash from Low-Sulfur and High-Sulfur Coal Sources: Partitioning of Carbon and Trace Elements with Particle Size. Energy Sources vol. 21, 511–525. https://doi.org/ 10.1080/00908319950014641.
- Hower, J., Robertson, J., Roberts, J., 2001. Petrology and minor element chemistry of combustion by-products from the co-combustion of coal, tire-derived fuel, and petroleum coke at a western Kentucky cyclone-fired unit. Fuel Process. Technol. vol. 74, 125–142. https://doi.org/10.1016/S0378-3820(01)00224-7.
- Hower, J., Robl, T., Anderson, C., Thomas, G., Sakulpitakphon, T., Mardon, S., Clarl, W., 2005. Characteristics of coal combustion products (CCP's) from Kentucky power plants, with emphasis on mercury content. Fuel vol. 84, 1338–1350. https://doi.org/ 10.1016/j.fuel.2004.09.029.
- Hower, J., Eble, C., O'Keefe, J., Dai, S., Wang, P., Xie, P., Liu, J., Ward, C., French, D., 2015. Petrology, Palynology, and Geochemistry of Gray Hawk Coal (Early Pennsylvanian, Langsettian) in Eastern Kentucky, USA. Minerals vol. 5, 592–622. https://doi.org/10.3390/min5030511.
- Hower, J., Granite, E., Mayfield, D., Lewis, A., Finkelman, R., 2016. Notes on Contributions to the Science of Rare Earth Element Enrichment in Coal and Coal Combustion Byproducts. Minerals vol. 6. https://doi.org/10.3390/min6020032.
- Hower, J., Groppo, J., Graham, U., Ward, C., Kostova, I., Maroto-Valer, M., Dai, S., 2017. Coal-derived unburned carbons in fly ash: A review. Int. J. Coal Geol. vol. 179, 11–27. https://doi.org/10.1016/j.coal.2017.05.007.
- Hower, J., Groppo, J., Joshi, P., Preda, D., Gamliel, D., Mohler, D., Wiseman, J., Hopps, S., Morgan, T., Beers, T., Schrock, M., 2020. Distribution of Lanthanides, Yttrium, and Scandium in the Pilot-Scale Beneficiation of Fly Ashes Derived from Eastern Kentucky Coals. Minerals vol. 10. https://doi.org/10.3390/min10020105.
- Huffman, G., Huggins, F., Shah, N., Huggins, R., Linak, W., Miller, C., Pugmire, R., Meuzelaar, H., Seehra, M., Manivannan, A., 2000. Characterization of Fine Particulate Matter Produced by Combustion of Residual Fuel Oil. J. Air Waste Manag. Assoc. vol. 50, 1106–1114. https://doi.org/10.1080/ 10473289.2000.10464157.
- Huggins, F., Huffman, G., 1996. Modes of occurrence of trace elements in coal from XAFS spectroscopy. Int. J. Coal Geol. 32, 31–53.
- Huggins, F., Huffman, G., 2004. How do lithophile elements occur in organic association in bituminous coals. Int. J. Coal Geol. vol. 58, 193–204. https://doi.org/10.1016/j. coal.2003.10.009.
- Huggins, F., Srikantapura, S., Parekh, B., Blanchard, L., Robertson, J., 1997. XANES Spectroscopic Characterization of Selected Elements in Deep-Cleaned Fractions of Kentucky No. 9 Coal. Energy Fuels vol. 11, 691–701. https://doi.org/10.1021/ ef960118u.
- Huggins, F., Seidu, L., Shah, N., Huffman, G., Honaker, R., Kyger, J., Higgins, B., Robertson, J., Pal, S., Seehra, M., 2009. Elemental modes of occurrence in an Illinois #6 coal and fractions prepared by physical separation techniques at a coal preparation plant. Int. J. Coal Geol. vol. 78, 65–76. https://doi.org/10.1016/j. coal.2008.10.002.
- Izquierdo, M., Querol, X., 2012. Leaching behaviour of elements from coal combustion fly ash: An overview. Int. J. Coal Geol. vol. 94, 54–66. https://doi.org/10.1016/j. coal.2011.10.006.
- Izquierdo, M., Font, O., Moreno, N., Querol, X., Huggins, F., Alvarez, E., Diez, S., Otero, P., Ballesteros, J., Gimenez, A., 2007. Influence of a Modification of the Petcoke/Coal Ratio on the Leachability of Fly Ash and Slag Produced from a Large PCC Power Plant. Environ. Sci. Technol. vol. 41, 5330–5335. https://doi.org/ 10.1021/es063002d.
- Jadhav, U., Hocheng, H., 2015. Analysis of Metal Bioleaching from Thermal Power Plant Fly Ash by Aspergillus niger 34770 Culture Supernatant and Reduction of Phytotoxicity During the Process. Appl. Biochem. Biotechnol. vol. 175, 870–881. https://doi.org/10.1007/s12010-014-1323-2.
- Jain, N., Sharma, D., 2004. Biohydrometallurgy for nonsulfidic minerals—a review. Geomicrobiol. J. vol. 21, 135–144. https://doi.org/10.1080/01490450490275271
- Jano-Ito, M., Reed, G., Millan, M., 2014. Comparison of thermodynamic equilibrium predictions on trace element speciation in oxy-fuel and conventional coal combustion power plants. Energy Fuels vol. 28, 4666–4683. https://doi.org/ 10.1021/ef5005607.
- Jia, L., Anthony, E., Charland, J., 2002. Investigation of Vanadium Compounds in Ashes from a CFBC Firing 100 Petroleum Coke. Energy Fuels vol. 16, 397–403. https://doi. org/10.1021/ef0102380.
- Jiang, P., Xie, C., Luo, C., Meng, W., Yang, G., Yu, G., Gong, Y., Xu, M., Wu, T., 2021. Distribution and modes of occurrence of heavy metals in opposed multi-burner coalwater-slurry gasification plants. Fuel vol. 303. https://doi.org/10.1016/j. fuel.2021.121163.
- Kantor, P., Raclavska, H., Matysek, D., Raclavsky, K., Svedova, B., Kucbel, M., 2021. Sources of Magnetic Particles from Air Pollution in Mountainous Area. Inżynieria Miner. vol. 1 https://doi.org/10.29227/IM-2019-01-08.

- Karayigit, A., Gayer, R., Querol, X., Onacak, T., 2000. Contents of major and trace elements in feed coals from Turkish coal-fired power plants. Int. J. Coal Geol. 44, 169–184.
- Karayigit, A., Bulut, Y., Karayigit, G., Querol, X., Alastuey, A., Vassilev, S., Vassileva, C., 2007. Mass Balance of Major and Trace Elements in a Coal-Fired Power Plant. Energy Sources, Part A: Recovery, Util., Environ. Eff. vol. 28, 1311–1320. https:// doi.org/10.1080/009083190910523.
- Kashiwakura, S., Ohno, H., Matsubae-Yokoyama, K., Kumagai, Y., Kubo, H., Nagasaka, T., 2010. Removal of arsenic in coal fly ash by acid washing process using dilute H2SO4 solvent. J. Hazard. Mater. vol. 181, 419–425. https://doi.org/ 10.1016/j.jhazmat.2010.05.027.
- Kashiwakura, S., Ohno, H., Kumagai, Y., Kubo, H., Matsubae, K., Nagasaka, T., 2011. Dissolution behavior of selenium from coal fly ash particles for the development of an acid-washing process. Chemosphere vol. 85, 598–602. https://doi.org/10.1016/j. chemosphere.2011.06.099.
- Kashiwakura, S., Kumagai, Y., Kubo, H., Wagatsuma, K., 2013. Dissolution of Rare Earth Elements from Coal Fly Ash Particles in a Dilute H2SO4 Solvent. Open J. Phys. Chem. vol. 03, 69–75. https://doi.org/10.4236/ojpc.2013.32009.
- Kermer, R., Hedrich, S., Bellenberg, S., Brett, B., Schrader, D., Schönherr, P., Köpcke, M., Siewert, K., Günther, N., Gehrke, T., Sand, W., Räuchle, K., Bertau, M., Heide, G., Weitkämper, L., Wotruba, H., Ludwig, H., Partusch, R., Schippers, A., Reichel, S., Glombitza, F., Janneck, E., 2017. Lignite ash: Waste material or potential resource -Investigation of metal recovery and utilization options. Hydrometallurgy vol. 168, 141–152. https://doi.org/10.1016/j.hydromet.2016.07.002.
- Ketris, M., Yudovich, Y., 2009. Estimations of Clarkes for Carbonaceous biolithes: World averages for trace element contents in black shales and coals. Int. J. Coal Geol. vol. 78, 135–148. https://doi.org/10.1016/j.coal.2009.01.002.
- Klavins, M., Silamikele, I., Nikodemus, O., Kalnina, L., Kuske, E., Rodinov, V., Purmalis, O., 2009. Peat properties, major and trace element accumulation in bog peat in Latvia. Baltica 22, 37–50.
- Klika, Z., Bartoňová, L., Spears, D., 2001. Effect of boiler output on trace element partitioning during coal combustion in two fluidised-bed power stations. Fuel vol. 80, 907–917. https://doi.org/10.1016/S0016-2361(00)00164-2.
- Kostova, I., Vassileva, C., Dai, S., Hower, J., 2016. Mineralogy, geochemistry and mercury content characterization of fly ashes from the Maritza 3 and Varna thermoelectric power plants, Bulgaria. Fuel vol. 186, 674–684. https://doi.org/ 10.1016/j.fuel.2016.09.015.
- Kucbel, M., Raclavská, H., Růžičková, J., Švédová, B., Sassmanová, V., Drozdová, J., Raclavský, K., Juchelková, D., 2019. Properties of composts from household food waste produced in automatic composters. J. Environ. Manag. vol. 236, 657–666. https://doi.org/10.1016/j.jenvman.2019.02.018.
- Lanzerstorfer, C., 2018. Fly ash from coal combustion: Dependence of the concentration of various elements on the particle size. Fuel vol. 228, 263–271. https://doi.org/ 10.1016/j.fuel.2018.04.136.
- Lee, J., Pandey, B., 2012. Bio-processing of solid wastes and secondary resources for metal extraction – A review. Waste Manag. vol. 32, 3–18. https://doi.org/10.1016/j. wasman.2011.08.010.
- Li, B., Zhuang, X., Querol, X., Li, J., Moreno, N., Córdoba, P., Shangguan, Y., Zhou, J., Ma, X., Liu, S., 2019. Geological controls on enrichment of Mn, Nb (Ta), Zr (Hf), and REY within the Early Permian coals of the Jimunai Depression, Xinjiang Province, NW China. Int. J. Coal Geol. vol. 215 https://doi.org/10.1016/j.coal.2019.103298.
- Li, B., Zhuang, X., Querol, X., Moreno, N., Córdoba, P., Shangguan, Y., Yang, L., Li, J., Zhang, F., 2020. Geological controls on the distribution of REY-Zr (Hf)-Nb (Ta) enrichment horizons in late Permian coals from the Qiandongbei Coalfield, Guizhou Province, SW China. Int. J. Coal Geol. vol. 231 https://doi.org/10.1016/j. coal.2020.103604.
- Li, H., Fang, H., Wang, K., Zhou, W., Yang, Z., Yan, X., Ge, W., Li, Q., Xie, B., 2015. Asynchronous extraction of vanadium and chromium from vanadium slag by stepwise sodium roasting-water leaching. Hydrometallurgy vol. 156, 124–135. https://doi.org/10.1016/j.hydromet.2015.06.003.
- Li, H., Wang, K., Hua, W., Yang, Z., Zhou, W., Xie, B., 2016. Selective leaching of vanadium in calcification-roasted vanadium slag by ammonium carbonate. Hydrometallurgy vol. 160, 18–25. https://doi.org/10.1016/j. hydromet.2015.11.014.
- Lieberman, N., Izquierdo, M., Muñoz-Quirós, C., Cohen, H., Chenery, S., 2020. Geochemical signature of superhigh organic sulphur Raša coals and the mobility of toxic trace elements from combustion products and polluted soils near the Plomin coal-fired power station in Croatia. Appl. Geochem. vol. 114 https://doi.org/ 10.1016/j.apgeochem.2019.104472.
- Liu, Y., Liu, G., Qi, C., Cheng, S., Sun, R., 2016. Chemical speciation and combustion behavior of chromium (Cr) and vanadium (V) in coals. Fuel vol. 184, 42–49. https:// doi.org/10.1016/j.fuel.2016.07.003.
- Liu, Y., Liu, G., Qu, Q., Qi, C., Sun, R., Liu, H., 2017. Geochemistry of vanadium (V) in Chinese coals. Environ. Geochem. Health vol. 39, 967–986. https://doi.org/ 10.1007/s10653-016-9877-2.
- Low, F., De Girolamo, A., Wu, X., Dai, B., Zhang, L., 2015. Inhibition of lignite ash slagging and fouling upon the use of a silica-based additive in an industrial pulverised coal-fired boiler: Part 3 – Partitioning of trace elements. Fuel vol. 139, 746–756. https://doi.org/10.1016/j.fuel.2014.09.015.
- Lv, B., Jiao, F., Chen, Z., Dong, B., Fang, C., Zhang, C., Deng, X., 2022. Separation of unburned carbon from coal fly ash: Pre-classification in liquid–solid fluidized beds and subsequent flotation. Process Saf. Environ. Prot. vol. 165, 408–419. https://doi. org/10.1016/j.psep.2022.07.031.
- Mardon, S., Hower, J., 2008. J.O'Keefe, M.Marks, D.Hedges, Coal combustion by-product quality at two stoker boilers: Coal source vs. fly ash collection system design. Int. J. Coal Geol. vol. 75, 248–254. https://doi.org/10.1016/j.coal.2008.07.004.

Masoum, H., Rastegar, S., Khamforoush, M., 2021. Ultrasound-assisted leaching of vanadium and yttrium from coal ash: optimization, kinetic and thermodynamic study. Chem. Eng. Technol. vol. 44, 2249–2256. https://doi.org/10.1002/ ceat.202100297.

- Matjie, R., Bunt, J., 2005. J.van Heerden, Extraction of alumina from coal fly ash generated from a selected low rank bituminous South African coal. Miner. Eng. 18, 299–310.
- Maylotte, D., Wong, J., St. Peters, R., Lytle, F., Greegor, R., 1981. X-ray absorption spectroscopic investigation of trace vanadium sites in coal. Science vol. 214, 554–556. https://doi.org/10.1126/science.214.4520.554.
- McIlveen-Wright, D., Huang, Y., Rezvani, S., Wang, Y., 2007. A technical and environmental analysis of co-combustion of coal and biomass in fluidised bed technologies. Fuel vol. 86, 2032–2042. https://doi.org/10.1016/j.fuel.2007.02.011.
- Meer, I., Nazir, R., 2018. Removal techniques for heavy metals from fly ash. J. Mater. Cycles Waste Manag. vol. 20, 703–722. https://doi.org/10.1007/s10163-017-0651z.
- Middleton, A., Park, D., Jiao, Y., Hsu-Kim, H., 2020. Major element composition controls rare earth element solubility during leaching of coal fly ash and coal by-products. Int. J. Coal Geol. vol. 227 https://doi.org/10.1016/j.coal.2020.103532.
- Mikulčić, H., von Berg, E., Vujanović, M., Wang, X., Tan, H., Duić, N., 2016. Numerical evaluation of different pulverized coal and solid recovered fuel co-firing modes inside a large-scale cement calciner. Appl. Energy vol. 184, 1292–1305. https://doi. org/10.1016/j.apenergy.2016.05.012.
- Mikulčić, H., Baleta, J., Klemeš, J., 2022. Cleaner technologies for sustainable development. Clean. Eng. Technol. vol. 7 https://doi.org/10.1016/j. clet.2022.100445.
- Mirazimi, S., Abbasalipour, Z., Rashchi, F., 2015. Vanadium removal from LD converter slag using bacteria and fungi. J. Environ. Manag. vol. 153, 144–151. https://doi.org/ 10.1016/j.jenvman.2015.02.008.
- Moskalyk, R., Alfantazi, A., 2003. Processing of vanadium: a review. Miner. Eng. vol. 16, 793–805. https://doi.org/10.1016/S0892-6875(03)00213-9.
- Mukherjee, A., Kaantee, U., Zevenhoven, R., 2003. The effects of switching from coal to alternative fuels on heavy metals from cement manufacturing., in: In: Chemistry of Trace elements in fly ash. 6th International Conference on the biochemistry of trace elements., Kluwer Academic/Plenum Publishers, New York, USA, 2003: pp. 45–61.
- Mukherjee, K., Dutta, N., Chandra, D., Singh, M., 1992. Geochemistry of trace elements of Tertiary coals of India. Int. J. Coal Geol. vol. 20, 99–113. https://doi.org/ 10.1016/0166-5162(92)90006-I.
- Myazin, V., Shumilova, L., Razmakhnin, K., Bogidaev, S., 2018. Integrated processing of ash and slag from thermal power plants in Eastern Transbaikalia. J. Min. Sci. vol. 54, 845–857. https://doi.org/10.1134/S1062739118054975.
- Narodoslawsky, M., Obernberger, I., 1996. From waste to raw material—the route from biomass to wood ash for cadmium and other heavy metals. J. Hazard. Mater. vol. 50, 157–168. https://doi.org/10.1016/0304-3894(96)01785-2.
- Navarro, R., Guzman, J., Saucedo, I., Revilla, J., Guibal, E., 2007. Vanadium recovery from oil fly ash by leaching, precipitation and solvent extraction processes. Waste Manag. vol. 27, 425–438. https://doi.org/10.1016/j.wasman.2006.02.002.
- Nazari, E., Rashchi, F., Saba, M., Mirazimi, S., 2014. Simultaneous recovery of vanadium and nickel from power plant fly-ash: Optimization of parameters using response surface methodology. Waste Manag. vol. 34, 2687–2696. https://doi.org/10.1016/j. wasman.2014.08.021.
- Nodelman, I., Pisupati, S., Miller, S., Scaroni, A., 2000. Partitioning behavior of trace elements during pilot-scale combustion of pulverized coal and coal–water slurry fuel. J. Hazard. Mater. vol. 74. 47–59. https://doi.org/10.1016/S0304-3894(99)00198-3.
- Nugteren, H., Janssen-Jurkovícová, M., Scarlett, B., 2002. Removal of heavy metals from fly ash and the impact on its quality. J. Chem. Technol. Biotechnol. vol. 77, 389–395. https://doi.org/10.1002/jctb.600.
- Nunes, M., Pereira, C., Guedes, A., Santos, A., Valentim, B., Freire, C., 2022. Assessment of coal fly ash char as a substituting material of graphite with electrocatalytic activity for the oxygen reduction reaction. Sustain. Chem. Pharm. vol. 27 https:// doi.org/10.1016/j.scp.2022.100705.
- doi.org/10.1016/j.scp.2022.100705.
 Oboirien, B., Thulari, V., North, B., 2014. Major and trace elements in coal bottom ash at different oxy coal combustion conditions. Appl. Energy vol. 129, 207–216. https://doi.org/10.1016/j.apenergy.2014.04.091.
- Oliveira, M., Ward, C., French, D., Hower, J., Querol, X., Silva, L., 2012. Mineralogy and leaching characteristics of beneficiated coal products from Santa Catarina, Brazil. Int. J. Coal Geol. vol. 94, 314–325. https://doi.org/10.1016/j.coal.2011.10.004.
- Ordorica-Garcia, G., Douglas, P., Croiset, E., Zheng, L., 2006. Technoeconomic evaluation of IGCC power plants for CO₂ avoidance. Energy Convers. Manag. vol. 47, 2250–2259. https://doi.org/10.1016/j.enconman.2005.11.020.
- Orem, W., Finkelman, R., 2003. Coal formation and geochemistry. Treatise on Geochemistry. Elsevier,, pp. 191–222. https://doi.org/10.1016/B0-08-043751-6/ 07097-3.
- Palmer, C., Tuncalı, E., Dennen, K., Coburn, T., Finkelman, R., 2004. Characterization of Turkish coals: a nationwide perspective. Int. J. Coal Geol. vol. 60, 85–115. https:// doi.org/10.1016/j.coal.2004.05.001.
- Park, S., Liang, Y., 2019. Bioleaching of trace elements and rare earth elements from coal fly ash. Int. J. Coal Sci. Technol. vol. 6, 74–83. https://doi.org/10.1007/s40789-019-0238-5.
- Parzentny, H., Róg, L., 2020. Distribution of some ecotoxic elements in fuel and solid combustion residues in Poland. Energies vol. 13. https://doi.org/10.3390/ en13051131.
- Paul, M., Seferinoğlu, M., Ayçık, G., Sandström, Å., Smith, M., Paul, J., 2006. Acid leaching of ash and coal: Time dependence and trace element occurrences. Int. J. Miner. Process. vol. 79, 27–41. https://doi.org/10.1016/j.minpro.2005.11.008.

- Pędziwiatr, A., Potysz, A., Uzarowicz, Ł., 2021. Combustion wastes from thermal power stations and household stoves: A comparison of properties, mineralogical and chemical composition, and element mobilization by water and fertilizers. Waste Manag. vol. 131, 136–146. https://doi.org/10.1016/j.wasman.2021.05.035.
- Peiravi, M., Ackah, L., Guru, R., Mohanty, M., Liu, J., Xu, B., Zhu, X., Chen, L., 2017. Chemical extraction of rare earth elements from coal ash. Miner. Metall. Process. vol. 34, 170–177. https://doi.org/10.19150/mmp.7856.
- Perämäki, S., 2014. Method development for determination and recovery of rare earth elements from industrial fly ash. University of Jyväskylä. Academic Dissertation for the Degree of Doctor of Philosophy., Finland.
- Pires, M., Querol, X., 2004. Characterization of Candiota (South Brazil) coal and combustion by-product. Int. J. Coal Geol. vol. 60, 57–72. https://doi.org/10.1016/j. coal.2004.04.003.
- Pollock, S., Goodarzi, F., Riegider, C., 2000. Mineralogical and elemental variation of coal from Alberta, Canada: an example from the No. 2 seam, Genesee Mine. Int. J. Coal Geol. 43, 259–586.
- Querol, X., Fernández-Turiel, J., López-Soler, A., 1995. trace elements in coal and their behaviour during combustion in a large power station. Fuel 74, 331–343.
- Querol, X., UmaÑa, J., Alastuey, A., Bertrana, C., Lopez-Soler, A., Plana, F., 2000. Extraction of Water-Soluble Impurities from Fly Ash. Energy Sources vol. 22, 733–749. https://doi.org/10.1080/00908310050120281.
- Querol, X., Alastuey, A., Zhuang, X., Hower, J., Lopez-Soler, A., Plana, F., Zeng, R., 2001. Petrology, mineralogy and goechemistry of the permian and Triassic coals in the leping area, Jiangxi Province, southeast China. Int. J. Coal Geol. 48, 23–45.
- Querol, X., Izquierdo, M., Monfort, E., Alvarez, E., Font, O., Moreno, T., Alastuey, A., Zhuang, X., Lu, W., Wang, Y., 2008. Environmental characterization of burnt coal gangue banks at Yangquan, Shanxi Province, China. Int. J. Coal Geol. vol. 75, 93-104. https://doi.org/10.1016/j.coal.2008.04.003.
- Quick, W., Irons, R., 2002. Trace element partitioning during the firing of washed and untreated power station coals. Fuel vol. 81, 665–672. https://doi.org/10.1016/ S0016-2361(01)00197-1.
- Raclavska, H., Juchelkova, D., Roubicek, V., Matysek, D., 2011. Energy utilisation of biowaste — Sunflower-seed hulls for co-firing with coal. Fuel Process. Technol. vol. 92, 13–20. https://doi.org/10.1016/j.fuproc.2010.03.006.
- Raclavská, H., Corsaro, A., Hlavsová, A., Juchelková, D., Zajonc, O., 2015. The effect of moisture on the release and enrichment of heavy metals during pyrolysis of municipal solid waste. Waste Manag. Res.: J. a Sustain. Circ. Econ. vol. 33, 267–274. https://doi.org/10.1177/0734242X14568609.
- Raclavská, H., Růžičková, J., Škrobánková, H., Koval, S., Kucbel, M., Raclavský, K., Švédová, B., Pavlík, P., Juchelková, D., 2018. Possibilities of the utilization of char from the pyrolysis of tetrapak. J. Environ. Manag. vol. 219, 231–238. https://doi. org/10.1016/j.jenvman.2018.05.002.
- Rastegar, S., Mousavi, S., Shojaosadati, S., Sarraf Mamoory, R., 2015. Bioleaching of V, Ni, and Cu from residual produced in oil fired furnaces using Acidithiobacillus ferrooxidans. Hydrometallurgy vol. 157, 50–59. https://doi.org/10.1016/j. hydromet.2015.07.006.
- Ren, D., Zhao, F., Dai, S., Zhang, J., Luo, K., 2006. Geochemistry of trace elements in coal.: (in Chinese with English abstract). Science Press,, Beijing.
- Riley, K., 2007. Bench-marking of Fly Ash. CCSD Research Report no 67. In: Cooperative Research Centre for Coal in Sustainable Development, Pullenwalle, QLD, Australia.
- Roy, B., Choo, W., Bhattacharya, S., 2013. Prediction of distribution of trace elements under Oxy-fuel combustion condition using Victorian brown coals. Fuel vol. 114, 135–142. https://doi.org/10.1016/i.fuel.2012.09.080.
- 135–142. https://doi.org/10.1016/j.fuel.2012.09.080.
 Růžičková, J., Raclavská, H., Kucbel, M., Raclavský, K., Šafář, M., Švédová, B., 2018. The possibilities to identify combusted fuel based on an analysis of ash from local heating. J. Environ. Manag. vol. 219, 269–276. https://doi.org/10.1016/j. ienvman.2018.05.001.
- Růžičková, J., Kucbel, M., Raclavská, H., Švédová, B., Raclavský, K., Šafář, M., Kantor, P., 2019a. Chemical and Mineralogical Composition of Soot and Ash from the Combustion of Peat Briquettes in Household Boilers. Energies vol. 12. https://doi. org/10.3390/en12193784.
- Růžičková, J., Kucbel, M., Raclavská, H., Švédová, B., Raclavský, K., Juchelková, D., 2019b. Comparison of organic compounds in char and soot from the combustion of biomass in boilers of various emission classes. J. Environ. Manag. vol. 236, 769–783. https://doi.org/10.1016/j.jenvman.2019.02.038.
- Sandeep, P., Sahu, S., Kothai, P., Pandit, G., 2016. Leaching Behavior of Selected Trace and Toxic Metals in Coal Fly Ash Samples Collected from Two Thermal Power Plants, India. Bull. Environ. Contam. Toxicol. vol. 97, 425–431. https://doi.org/10.1007/ s00128-016-1864-x.
- Santos, A., Guedes, A., French, D., Futuro, A., Valentim, B., 2022. Integrative Study Assessing Space and Time Variations with Emphasis on Rare Earth Element (REE) Distribution and Their Potential on Ashes from Commercial (Colombian) Coal. Minerals vol. 12. https://doi.org/10.3390/min12020194.
- Scala, F., Chirone, R., Lancia, A., 2011. Elemental mercury vapor capture by powdered activated carbon in a fluidized bed reactor. Fuel vol. 90, 2077–2082. https://doi. org/10.1016/j.fuel.2011.02.042.
- Seferinoglu, M., 2003. Acid leaching of coal and coal-ashes*. Fuel vol. 82, 1721–1734. https://doi.org/10.1016/S0016-2361(03)00132-7.
- Seidel, A., Zimmels, Y., 1998. Mechanism and kinetics of aluminum and iron leaching from coal fly ash by sulfuric acid. Chem. Eng. Sci. vol. 53, 3835–3852. https://doi. org/10.1016/S0009-2509(98)00201-2.
- Senneca, O., Scala, F., Chirone, R., Salatino, P., 2017. Relevance of structure, fragmentation and reactivity of coal to combustion and oxy-combustion. Fuel vol. 201, 65–80. https://doi.org/10.1016/j.fuel.2016.11.034.
- Shabtai, J., Fleminger, G., Fleming, J., 1993, Extraction of metal oxides from coal fly ash by microorganisms and a new microorganism useful therefore. US Patent 5,231,018.

Silva, L., Ward, C., Hower, J., Izquierdo, M., Waanders, F., Oliveira, M., Li, Z., Hatch, R., Querol, X., 2010. Mineralogy and Leaching Characteristics of Coal Ash from a Major Brazilian Power Plant. Coal Combust. Gasif. Prod. vol. 2, 51–65. https://doi.org/ 10.4177/CCGP-D-10-00005.1.

- Silva, L., Oliveira, M., Sampaio, C., de Brum, I., Hower, J., 2013. Vanadium and nickel speciation in pulverized coal and petroleum coke co-combustion. Energy Fuels vol. 27, 1194–1203. https://doi.org/10.1021/ef4000038.
- Silva, L., Oliveira, M., Kautzmann, R., Ramos, C., Izquierdo, M., Dai, S., Wilcox, J., Hoffman, J., Hower, J., 2014. Geochemistry and mineralogy of coal-fired circulating fluidised bed combustion fly ashes. Coal Combust. Gasif. Prod. 6, 6–28. https://doi. org/10.4177/CCGP-D-14.00005.1.
- Singer, A., Navrot, J., Shapira, R., 1982. Extraction of aluminum from fly-ash by commercial and microbiologically-produced citric acid. Eur. J. Appl. Microbiol. Biotechnol. vol. 16, 228–230. https://doi.org/10.1007/BF00505838.
- Singh, S., Ram, L., Masto, R., Verma, S., 2011. A comparative evaluation of minerals and trace elements in the ashes from lignite, coal refuse, and biomass fired power plants. Int. J. Coal Geol. vol. 87, 112–120. https://doi.org/10.1016/j.coal.2011.05.006.
- Sloss, L., 2007. Trace elements and fly ash utilization. IEA Clean Coal Centre, Technical report., in: 2007.
- Snegrev, V., Sabirova, T., 2021. State-of-the-art and problems of bioleaching of metals from ash-and-slag wastes. Metallurgist vol. 65, 794–807. https://doi.org/10.1007/ s11015-021-01217-7.
- Spears, A.D., 2013. The determination of trace element distributions in coals using sequential chemical leaching—A new approach to an old method. Fuel vol. 114, 31–37. https://doi.org/10.1016/j.fuel.2012.09.028.
- Spears, A.D., 2015. The geochemistry and mineralogy of high-S coals, with examples mainly from the East Pennines Coalfield in Yorkshire and Nottinghamshire, UK: an overview. Proceedings of the Yorkshire Geological Society 60, 204–226. https://doi. org/10.1144/pygs2015–356.
- Spears, D., 2017. The role of seawater on the trace element geochemistry of some UK coals and a tribute to goldschmidt. Minerals vol. 7. https://doi.org/10.3390/ min7080148.
- Spears, D., Martinez-Tarrazona, M., 2004. Trace elements in combustion residues from a UK power station. Fuel vol. 83, 2265–2270. https://doi.org/10.1016/j. fuel.2004.06.025.
- Spears, D., Zheng, Y., 1999. geochemistry and origin of elements in some UK coals. Int. J. Coal Geol. 38, 161–179.
- Stefaniak, S., Kmiecik, E., Miszczak, E., Szczepańska-Plewa, J., Twardowska, I., 2018. Leaching behavior of fly ash from co-firing of coal with alternative off gas fuel in powerplant boilers. Appl. Geochem. vol. 93, 129–144. https://doi.org/10.1016/j. apgeochem.2018.04.003.
- Strzałkowska, E., 2021. Morphology, chemical and mineralogical composition of magnetic fraction of coal fly ash. Int. J. Coal Geol. vol. 240 https://doi.org/10.1016/ j.coal.2021.103746.
- Su, H., Chen, H., Lin, J., 2020. A sequential integration approach using Aspergillus Niger to intensify coal fly ash as a rare metal pool. Fuel vol. 270. https://doi.org/10.1016/ j.fuel.2020.117460.
- Švédová, B., Kucbel, M., Raclavská, H., Růžičková, J., Raclavský, K., Sassmanová, V., 2019. Water-soluble ions in dust particles depending on meteorological conditions in urban environment. J. Environ. Manag. vol. 237, 322–331. https://doi.org/ 10.1016/j.jenvman.2019.02.086.
- Swaine, D., 1990. Trace Elements in Coal. Butterworth-Heinemann, https://doi.org/ 10.1016/C2013-0-00949-8.
- Swaine, D., 2000. Why trace elements are important. Fuel Process. Technol. 65–66, 21–33. https://doi.org/10.1016/S0378-3820(99)00073-9.
- Swaine, D., Goodarzi, F. (Eds.), 1995. Environmental Aspects of Trace Elements in Coal. Springer, Netherlands, Dordrecht. https://doi.org/10.1007/978-94-015-8496-8.
- Springer, Netherlands, Dordrecht. https://doi.org/10.1007/978-94-015-8496-8.
 Świetlik, R., Trojanowska, M., Jóźwiak, M., 2012. Evaluation of the distribution of heavy metals and their chemical forms in ESP-fractions of fly ash. Fuel Process. Technol. vol. 95, 109–118. https://doi.org/10.1016/j.fuproc.2011.11.019.
- Tang, Q., Liu, G., Zhou, C., Sun, R., 2013. Distribution of trace elements in feed coal and combustion residues from two coal-fired power plants at Huainan, Anhui, China. Fuel vol. 107, 315–322. https://doi.org/10.1016/j.fuel.2013.01.009.
- Tang, X., Huang, W., 2004. Trace Elem. Chin. Coals (Chin.).
 Torma, A., Singh, A., 1993. Acidolysis of coal fly ash by Aspergillus niger. Fuel vol. 72, 1625–1630. https://doi.org/10.1016/0016-2361(93)90346-4.
- Tsuboi, I., Kasai, S., Kunugita, E., Komasawa, I., 1991. Recovery of gallium and vanadium from coal fly ash. J. Chem. Eng. Jpn. vol. 24, 15–20. https://doi.org/ 10.1252/jcej.24.15.
- Tun, M., Juchelková, D., 2019a. Biomass Sources and Energy Potential for Energy Sector in Myanmar: An Outlook. Resources vol. 8. https://doi.org/10.3390/ resources8020102.
- Tun, M., Juchelková, D., 2019b. Drying methods for municipal solid waste quality improvement in the developed and developing countries: A review. Environ. Eng. Res. vol. 24, 529–542. https://doi.org/10.4491/eer.2018.327.
- Tun, M., Juchelková, D., Raclavská, H., Sassmanová, V., 2018. Utilization of Biodegradable Wastes as a Clean Energy Source in the Developing Countries: A Case Study in Myanmar. Energies vol. 11. https://doi.org/10.3390/en11113183.
- US EPA, Pollutants in Each Pollutant Group Sorted by Group Name.: Listing of Pollutants in each RBLC Pollutant Group, 2011. https://www3.epa.gov/dirl/poltgrps_eg.
- Valeev, D., Bobylev, P., Osokin, N., Zolotova, I., Rodionov, I., Salazar-Concha, C., Verichev, K., 2022. A review of the alumina production from coal fly ash, with a focus in Russia. J. Clean. Prod. vol. 363 https://doi.org/10.1016/j. jclepro.2022.132360.
- Valentim, B., Abagiu, A., Anghelescu, L., Flores, D., French, D., Gonçalves, P., Guedes, A., Popescu, L., Predeanu, G., Ribeiro, J., Santos, A., Slåvescu, V., Ward, C., 2019.

Assessment of bottom ash landfilled at Ceplea Valley (Romania) as a source of rare earth elements. Int. J. Coal Geol. vol. 201, 109–126. https://doi.org/10.1016/j. coal.2018.11.019.

- Vassilev, S., Menendez, R., 2005. Phase-mineral and chemical composition of coal fly ashes as a basis for their multicomponent utilization. 4. Characterization of heavy concentrates and improved fly ash residues. Fuel vol. 84, 973–991. https://doi.org/ 10.1016/j.fuel.2004.11.021.
- Vassilev, S., Vassileva, C., 2020. Contents and associations of rare earth elements and yttrium in biomass ashes. Fuel vol. 262. https://doi.org/10.1016/j. fuel.2019.116525.
- Vassilev, S., Yossifova, M., Vassileva, C., 1994. Mineralogy and geochemistry of Bobov Dol coals, Bulgaria. Int. J. Coal Geol. vol. 26, 185–213. https://doi.org/10.1016/ 0166-5162(94)90010-8.
- Vassilev, S., Menendez, R., Borrego, A., Diaz-Somoano, M., 2004. M.Rosa Martinez-Tarazona, Phase-mineral and chemical composition of coal fly ashes as a basis for their multicomponent utilization. 3. Characterization of magnetic and char concentrates. Fuel vol. 83, 1563–1583. https://doi.org/10.1016/j.fuel.2004.01.010.
- Vassilev, S., Vassileva, C., Karayigit, A., Bulut, Y., Alastuey, A., Querol, X., 2005. Phasemineral and chemical composition of fractions separated from composite fly ashes at the Soma power station, Turkey. Int. J. Coal Geol. vol. 61, 65–85. https://doi.org/ 10.1016/j.coal.2004.05.004.
- Vassilev, S., Baxter, D., Andersen, L., Vassileva, C., 2010. An overview of the chemical composition of biomass. Fuel vol. 89, 913–933. https://doi.org/10.1016/j. fuel.2009.10.022.
- Vassilev, S., Vassileva, C., Baxter, D., 2014. Trace element concentrations and associations in some biomass ashes. Fuel vol. 129, 292–313. https://doi.org/ 10.1016/j.fuel.2014.04.001.
- Vassilev, S., Vassileva, C., Vassilev, V., 2015. Advantages and disadvantages of composition and properties of biomass in comparison with coal: An overview. Fuel vol. 158, 330–350. https://doi.org/10.1016/j.fuel.2015.05.050.
- Vassilev, S., Kossev, K., Vassileva, C., 2020. Trace elements in water-soluble fractions from Bulgarian coal fly ashes and their technological and environmental importance. Energy Fuels vol. 34, 13782–13798. https://doi.org/10.1021/acs. energyfuels.0c02537.
- Vershinina, K., Egorov, R., Strizhak, P., 2016. The ignition parameters of the coal-water slurry droplets at the different methods of injection into the hot oxidant flow. Appl. Therm. Eng. vol. 107, 10–20. https://doi.org/10.1016/j. applthermaleng.2016.06.156.
- Vershinina, K., Dorokhov, V., Romanov, D., Strizhak, P., 2020. Comparing the ignition parameters of promising coal fuels. Process Saf. Environ. Prot. vol. 139, 273–282. https://doi.org/10.1016/j.psep.2020.04.027.
- Vershinina, K., Dorokhov, V., Romanov, D., Strizhak, P., 2022. Combustion stages of waste-derived blends burned as pellets, layers, and droplets of slurry. Energy vol. 251. https://doi.org/10.1016/j.energy.2022.123897.
- Vitolo, S., Seggiani, M., Filippi, S., Brocchini, C., 2000. Recovery of vanadium from heavy oil and Orimulsion fly ashes. Hydrometallurgy vol. 57, 141–149. https://doi. org/10.1016/S0304-386X(00)00099-2.
- Wagner, N., Hlatshwayo, B., 2005. The occurrence of potentially hazardous trace elements in five Highveld coals, South Africa. Int. J. Coal Geol. vol. 63, 228–246. https://doi.org/10.1016/j.coal.2005.02.014.
- Wagner, N., Matiane, A., 2018. Rare earth elements in select Main Karoo Basin (South Africa) coal and coal ash samples. Int. J. Coal Geol. vol. 196, 82–92. https://doi.org/ 10.1016/j.coal.2018.06.020.
- Wagner, N., Tlotleng, M., 2012. Distribution of selected trace elements in density fractionated Waterberg coals from South Africa. Int. J. Coal Geol. vol. 94, 225–237. https://doi.org/10.1016/j.coal.2012.01.005.
- Wagner, N., Mokwena, M., Kolker, A., 2021. Occurrence and probable source of chromium enrichment in Permian coals, South Africa. South Afr. J. Geol. vol. 124, 611–626. https://doi.org/10.25131/sajg.124.0013.
- Wang, J., Takaya, A., Tomita, A., 2004. Leaching of ashes and chars for examining transformations of trace elements during coal combustion and pyrolysis. Fuel vol. 83, 651–660. https://doi.org/10.1016/j.fuel.2003.06.003.
- Ward, C., Spears, D., Booth, C., Staton, I., Gurba, L., 1999. Mineral matter and trace elements in coals of the Gunnedah Basin, New South Wales, Australia. Int. J. Coal Geol. vol. 40, 281–308. https://doi.org/10.1016/S0166-5162(99)00006-3.
- Wasielewski, R., Wojtasek, M., Plis, A., 2020. Investig. Fly. ash Co. -Combust. Altern. fuel (SRF) hard coal a stoker Boil. 46, 58–67. https://doi.org/10.24425/ aep.2020.133475.
- Wei, C., Cheng, S., Zhu, F., Tan, X., Li, W., Zhang, P., Miao, S., 2018. Digesting highaluminum coal fly ash with concentrated sulfuric acid at high temperatures. Hydrometallurgy vol. 180, 41–48. https://doi.org/10.1016/j. hydromet.2018.07.004.
- Wierońska, F., Burmistrz, P., Strugała, A., Makowska, D., Lech, S., 2018. Effect of Using Coke Dust as a Sorbent for Removing Mercury from Flue Gases on the Contents of Selected Ecotoxic Elements in Fly Ash. Energy Fuels vol. 32, 5693–5700. https://doi. org/10.1021/acs.energyfuels.7b03557.
- Wierońska-Wiśniewska, F., Makowska, D., Strugała, A., 2022. Arsenic in polish coals: Content, mode of occurrence, and distribution during coal combustion process. Fuel vol. 312. https://doi.org/10.1016/j.fuel.2021.122992.
- Wojtaszek, M., Wasielewski, R., Kalaitzidis, S., 2021. Organic Petrographical Features of Fly Ashes Originating from Coal and Coal-SRF Co-Combustion. Minerals vol. 11. https://doi.org/10.3390/min11020128.
- Wu, P., Li, J., Zhuang, X., Querol, X., Moreno, N., Li, B., Ge, D., Zhao, S., Ma, X., Cordoba, P., Shangguan, Y., 2019. Mineralogical and Environmental Geochemistry of Coal Combustion Products from Shenhuo and Yihua Power Plants in Xinjiang

Autonomous Region, Northwest China. Minerals vol. 9. https://doi.org/10.3390/min9080496.

- Xu, J., Huang, Q., Wang, F., 2020. Co-combustion of municipal solid waste and coal for carbon emission reduction: A bi-level multi-objective programming approach.
 J. Clean. Prod. vol. 272 https://doi.org/10.1016/j.jclepro.2020.121923.
- Xu, Y., Wang, H., Liu, X., Zhu, J., Xu, J., Xu, M., 2022. Mitigating CO₂ emission in pulverized coal-fired power plant via co-firing ammonia: A simulation study of flue gas streams and exergy efficiency. Energy Convers. Manag. vol. 256 https://doi.org/ 10.1016/j.encomman.2022.115328.
- Yan, R., Gauthier, D., Flamant, G., 2001a. Volatility and chemistry of trace elements in a coal combustor. Fuel vol. 80, 2217–2226. https://doi.org/10.1016/S0016-2361(01) 00105-3.
- Yan, R., Gauthier, D., Flamant, G., 2001b. Partitioning of trace elements in the flue gas from coal combustion. Combust. Flame vol. 125, 942–954. https://doi.org/10.1016/ S0010-2180(00)00239-X.
- Yang, Z., Zhang, W., Li, Y., Wang, C., Yang, S., Tang, Z., Liu, D., Li, Y., 2020. Waterleaching characteristic of valuable trace metals (U, V, and Ga) from (NH4)2SO4treated coal ash: A coprecipitation behavior at high temperature. J. Hazard. Mater. vol. 388 https://doi.org/10.1016/j.jhazmat.2020.122113.
- Ye, P., Wang, X., Wang, M., Fan, Y., Xiang, X., 2012. Recovery of vanadium from stone coal acid leaching solution by coprecipitation, alkaline roasting and water leaching. Hydrometallurgy 117- 118, 108–115. https://doi.org/10.1016/j. hydromet.2012.02.013.
- Yermagambet, B., Nurgaliyev, N., Abylgazina, L., Maslov, N., Kassenova, Z., Kasenov, B., 2018. Methods for Extraction of Valuable Components from Ash-and-Slag Coal Wastes. Ser. Chem. Technol. vol. 6, 67–78. https://doi.org/10.32014/2018.2518-1491.28.
- Yossifova, M., 2014. Petrography, mineralogy and geochemistry of Balkan coals and their waste products. Int. J. Coal Geol. vol. 122, 1–20. https://doi.org/10.1016/j. coal.2013.12.007.
- Yu, M., Huang, Y., Xia, W., Fan, C., Zhu, Z., Li, Z., Zha, J., Xu, W., Yuan, Q., Cheng, H., 2022. Improved removal of selenium by calcite-derived CaO with porosity and antisintering in presence of acid gas. J. Environ. Chem. Eng. vol. 10 https://doi.org/ 10.1016/j.jece.2022.107342.
- Yudovich, Y., 2003. Coal inclusions in sedimentary rocks: a geochemical phenomenon. A review. Int. J. Coal Geol. vol. 56, 203–222. https://doi.org/10.1016/j. coal.2003.08.002.
- Zha, J., Huang, Y., Xia, W., Xia, Z., Liu, C., Dong, L., Liu, L., 2018. Effect of mineral reaction between calcium and aluminosilicate on heavy metal behavior during sludge incineration. Fuel vol. 229, 241–247. https://doi.org/10.1016/j. fuel.2018.05.015.
- Zha, J., Huang, Y., Clough, P., Dong, L., Xu, L., Liu, L., Zhu, Z., Yu, M., 2020. Desulfurization using limestone during sludge incineration in a fluidized bed furnace: Increased risk of particulate matter and heavy metal emissions. Fuel vol. 273. https://doi.org/10.1016/j.fuel.2020.117614.

- Zhang, S., Dai, S., Finkelman, R., Graham, I., French, D., Hower, J., Li, X., 2019b. Leaching characteristics of alkaline coal combustion by-products: A case study from a coal-fired power plant, Hebei Province, China. Fuel vol. 255. https://doi.org/ 10.1016/j.fuel.2019.115710.
- Zhang, Y., Zhang, T., Dreisinger, D., Lv, C., Lv, G., Zhang, W., 2019a. Recovery of vanadium from calcification roasted-acid leaching tailing by enhanced acid leaching. J. Hazard. Mater. vol. 369, 632–641. https://doi.org/10.1016/j. jhazmat.2019.02.081.
- Zhao, L., Dai, S., Finkelman, R., French, D., Graham, I., Yang, Y., Li, J., Yang, P., 2020. Leaching behavior of trace elements from fly ashes of five Chinese coal power plants. Int. J. Coal Geol. vol. 219 https://doi.org/10.1016/j.coal.2019.103381.
- Zhao, S., Duan, Y., Chen, L., Li, Y., Yao, T., Liu, S., Liu, M., Lu, J., 2017. Study on emission of hazardous trace elements in a 350 MW coal-fired power plant. Part 2. arsenic, chromium, barium, manganese, lead. Environ. Pollut. vol. 226, 404–411. https://doi.org/10.1016/j.envpol.2017.04.009.
- Zhao, S., Duan, Y., Lu, J., Gupta, R., Pudasainee, D., Liu, S., Liu, M., Lu, J., 2018a. Chemical speciation and leaching characteristics of hazardous trace elements in coal and fly ash from coal-fired power plants. Fuel vol. 232, 463–469. https://doi.org/ 10.1016/j.fuel.2018.05.135.
- Zhao, S., Duan, Y., Lu, J., Liu, S., Pudasainee, D., Gupta, R., Liu, M., Lu, J., 2018b. Enrichment characteristics, thermal stability and volatility of hazardous trace elements in fly ash from a coal-fired power plant. Fuel vol. 225, 490–498. https:// doi.org/10.1016/j.fuel.2018.03.190.
- Zhou, C., Liu, G., Cheng, S., Fang, T., Lam, P., 2014a. Thermochemical and trace element behavior of coal gangue, agricultural biomass and their blends during cocombustion. Bioresour. Technol. vol. 166, 243–251. https://doi.org/10.1016/j. biortech.2014.05.076.
- Zhou, C., Liu, G., Wu, D., Fang, T., Wang, R., Fan, X., 2014b. Mobility behavior and environmental implications of trace elements associated with coal gangue: A case study at the Huainan Coalfield in China. Chemosphere vol. 95, 193–199. https://doi. org/10.1016/j.chemosphere.2013.08.065.
- Zhou, C., Liu, G., Fang, T., Lam, P., 2015. Investigation on thermal and trace element characteristics during co-combustion biomass with coal gangue. Bioresour. Technol. vol. 175, 454–462. https://doi.org/10.1016/j.biortech.2014.10.129.
- Zhou, C., Liu, G., Wang, X., Qi, C., 2016. Co-combustion of bituminous coal and biomass fuel blends: Thermochemical characterization, potential utilization and environmental advantage. Bioresour. Technol. vol. 218, 418–427. https://doi.org/ 10.1016/j.biortech.2016.06.134.
- Zhou, C., Gao, F., Yu, Y., Zhang, W., Liu, G., 2020. Effects of gaseous agents on trace element emission behavior during co-combustion of coal with biomass. Energy Fuels vol. 34, 3843–3849. https://doi.org/10.1021/acs.energyfuels.9b04520.
- Zhou, C., Du, J., Zhang, Y., Sun, J., Wu, W., Liu, G., 2021. Redistribution and transformation mechanisms of gallium and germanium during coal combustion. Fuel vol. 305. https://doi.org/10.1016/j.fuel.2021.121532.